

# Walden University ScholarWorks

Walden Dissertations and Doctoral Studies

Walden Dissertations and Doctoral Studies Collection

2021

# Exploring knowledge retention strategies to prevent knowledge loss in project-based organizations (PBOs)

Robert Haughton Walden University

Follow this and additional works at: https://scholarworks.waldenu.edu/dissertations

# Walden University

College of Management and Technology

This is to certify that the doctoral dissertation by

# Robert Haughton

has been found to be complete and satisfactory in all respects, and that any and all revisions required by the review committee have been made.

**Review Committee** 

Dr. Steven Tippins, Committee Chairperson, Management Faculty Dr. Carol Wells, Committee Member, Management Faculty Dr. Nikunja Swain, University Reviewer, Management Faculty

Chief Academic Officer and Provost Sue Subocz, Ph.D.

Walden University 2021

# Abstract

Exploring Knowledge Retention Strategies to Prevent Knowledge Loss in Project-Based
Organizations (PBOs)

by

Robert Haughton

MS, Keller Graduate School of Management, 2005

BS, University of South Florida, 1993

Dissertation Submitted in Partial Fulfillment
of the Requirements for the Degree of
Doctor of Philosophy
Management

Walden University

May 2021

#### **Abstract**

Knowledge is a valuable asset, and managing that knowledge is now recognized as a significant contributor to organizations in the current business climate. The problem is that the loss of knowledge impacts project quality, organizational efficiency, and customer satisfaction. Too often, managers do not adequately communicate the knowledge retention strategies needed to reduce the impact of knowledge loss on project productivity. The purpose of this qualitative case study was to explore knowledge retention strategies to prevent knowledge loss in project-based organizations. The knowledge-based view of the firm and intellectual capital formed the conceptual framework. Eight participants consisting of managers, full-time employees, and consultants participated in the study. Data collection included semi-structured interviews using Zoom video teleconferencing software. Data analysis involved using two cycles of descriptive, pattern and focused coding to develop emerging codes and themes. This study revealed three themes: knowledge retention benefits, knowledge retention challenges, and knowledge retention strategies. This study's findings may help management develop and implement a knowledge retention strategy to identify, capture, and retain critical knowledge in their organization. The implication for positive social change may include organizations creating a positive workplace that influences employees to share and retain knowledge, increasing an organization's knowledge repository.

# Exploring Knowledge Retention Strategies to Prevent Knowledge Loss in Project-Based Organizations (PBOs)

by

# Robert Haughton

MS, Keller Graduate School of Management, 2005 BS, University of South Florida, 1993

Dissertation Submitted in Partial Fulfillment
of the Requirements for the Degree of
Doctor of Philosophy
Management

Walden University

May 2021

# Dedication

I dedicate this dissertation to all those who encouraged me to pursue my dream of earning a Ph.D. I appreciate all of the encouragement and motivation. A special feeling of gratitude to my wife, Jermaine, who supported and encouraged me throughout this process, for understanding the effort needed to complete this study, and for providing suggestions along the way. Without your support, I would not have finished this dissertation and achieved this major accomplishment in my life.

# Acknowledgments

I wanted to work on this topic because I saw what happens to project teams when people leave unexpectedly and take their knowledge with them. I saw team members struggle to pick up the slack on the project because either the knowledge transfer was unsuccessful or there was a lack of existing organizational knowledge on hand to help them finish the outstanding project tasks. This results in a decrease in project productivity and a decline in team morale. Some may consider this normal business as usual, but it should not be this way.

I would like to acknowledge all the people who played a role in the completion of this research. My sincere gratitude to my doctoral mentor and chair, Dr. Steven Tippins, who provided guidance and patience throughout this dissertation journey. I would also like to thank my other committee member, Dr. Anthony Lolas, and my URR, Dr. Nikunja Swain for the encouragement, guidance, and constructive feedback that helped me along the way to become a better scholar-practitioner and writer. I would like to thank all the research participants; thank you for sharing your thoughts and comments. To all my friends and former colleagues, who encouraged and kept pushing me along the way, a heartfelt thank you!

This dissertation journey has been hard, and no one said this would be easy.

Thank you to my Heavenly Father for giving me comfort and the strength to persevere when times were stressful and I wanted to throw in the towel.

# **Table of Contents**

List of Tables	۷i
Chapter 1: Introduction to the Study	.1
Background of the Study	2
Problem Statement	4
Purpose of the Study	4
Research Questions	.5
Conceptual Framework	.5
Nature of the Study	.7
Definitions	9
Assumptions1	0
Scope and Delimitations	1
Limitations1	1
Significance of the Study1	2
Significance to Business Practice	3
Significance to Theory	3
Significance to Social Change	4
Summary1	4
Chapter 2: Literature Review	6
Literature Search Strategy1	7
Conceptual Framework	8

Knowledge-Based View of the Firm	18	
Intellectual Capital	19	
What Is Knowledge?		
Tacit Knowledge	23	
Explicit Knowledge	24	
Knowledge Management	24	
Knowledge Creation	26	
Knowledge Sharing	27	
Knowledge Governance	29	
Organizational Learning	30	
Absorptive Capacity	33	
Project-Based Organizations	34	
Knowledge Assets	36	
Human Capital	37	
Social Capital	38	
Knowledge Retention	40	
Knowledge Loss	42	
Organizational Factors That Can Impact Knowledge Loss	43	
Employee Turnover	44	
Organizational Culture	46	
Leadership	47	

Summary and Conclusions	48
Chapter 3: Research Method	51
Research Design and Rationale	51
Role of the Researcher	55
Methodology	56
Participation Selection Logic	57
Instrumentation	59
Pilot Study	61
Data Collection	62
Data Analysis	63
Issues of Trustworthiness	65
Credibility	66
Transferability	67
Dependability	68
Confirmability	69
Ethical Procedures	69
Summary	70
Chapter 4: Results	72
Pilot Study	72
Research Setting	73
Demographics	73

	Data Collection	/5
	Data Analysis	76
	Evidence of Trustworthiness	79
	Credibility	79
	Transferability	80
	Dependability	80
	Confirmability	81
	Study Results	82
	Theme 1: Knowledge Retention Benefits	82
	Theme 2: Knowledge Retention Challenges	85
	Theme 3: Knowledge Retention Strategies	87
	Summary	90
Ch	apter 5: Discussion, Conclusions, and Recommendations	91
	Interpretation of Findings	91
	Limitations of the Study	95
	Recommendations	95
	Implications	97
	Contribution to Individuals	97
	Contribution to Business Practice	98
	Contributions to Society	99
	Contribution to Theory	100

References	103
Appendix A: Interview Questions – Managers	147
Appendix B: Interview Questions – Non-managers	149
Appendix C: Email Invitation to Potential Participants	151

# List of Tables

Table 1. Participant Profiles	74
Table 2. Themes and Codes	78
Table 3. Theme 1: Knowledge Retention Benefits	83
Table 4. Theme 2: Knowledge Retention Challenges	85
Table 5. Theme 3: Knowledge Retention Strategies	87

# Chapter 1: Introduction to the Study

Knowledge is a key component providing organizations with a competitive advantage in the current business environment (Fong & Kwok, 2009; Lin et al., 2016). Arsenijevic et al. (2017) asserted that knowledge is now a key source of wealth to many organizations. Therefore, it is important for organizations to identify, capture, and retain valuable knowledge or it will be lost. Companies in different industries are now using project management as their main strategy to deliver products and services to their customers. Miterev et al. (2017) noted that project-based organizations (PBOs) are able to adapt to changing customer demands and integrate knowledge into their project teams and organization.

A key to the success of PBOs are employees, specifically the knowledge they create, store, and share, and the relationships they develop play a pivotal role in delivering products and services to customers (Swart & Kinnie, 2013). However, challenges such as retaining valuable knowledge after the project closes and employee turnover occurs can lead to knowledge loss (Ali et al., 2018; Pee et al., 2014). Therefore, it is important that management in PBOs is proactive in mitigating these challenges and others to prevent project delays and knowledge loss.

This research study explored knowledge retention strategies to prevent knowledge loss in PBOs. The findings from this study may help managers to develop and implement a knowledge governance framework to identify, capture, and retain knowledge to strengthen their competitive advantage and achieve business objectives. Chapter 1

includes the background of the problem; problem statement; purpose of the study; research questions; conceptual framework; nature of the study; definition of the terms; assumptions; scope and delimitations; limitations; significance of the study to practice, theory, and social change; and a summary of the chapter.

# **Background of the Study**

Knowledge is recognized as the most significant asset for those companies that focus on capabilities and intangible resources as tools to compete in the marketplace; however, it is an asset not managed effectively (Whelan & Carcary, 2011). The management of knowledge is becoming more crucial because a majority of organizational, economic, and social activities are knowledge driven (Indira et al., 2012). Daghfous et al. (2013) stressed the importance of knowledge management as organizations realize that using knowledge resources effectively gives them the ability to be innovative, quickly respond to customer demands, and support operational activities. Massingham and Massingham (2014) indicated that companies in the United States invest in knowledge management by spending \$73 billion annually on software related to the concept alone; however, they also lose approximately \$31.5 billion a year by failing to share knowledge.

Many organizations are starting to organize their work on a project basis in response to rapid market changes and increasing customer demands (Almeida & Soares, 2014; Pemsel et al., 2014). The PBO has become an accepted business strategy and is able to create new organizational structures based on each project or customer demand

(Bourouni et al., 2015). A majority of PBOs perform several projects at the same time. Some of these projects are generally high risk, large, and complex and must be completed within a defined timeline and budget (Ajmal et al., 2010). In the current global economy that is categorized by intense competition and radically shrinking life cycles, projects have become an important vehicle for organizations to deliver products and services to their customers (Artto et al., 2015).

Within a PBO, management may move resources from other projects to meet the customer demands of another business-critical project, especially if it is behind schedule. This strategy is used when organizations do not have the needed specialized resources or have difficulty recruiting and hiring new employees with the necessary skillsets (Yaghootkar & Gil, 2012). The practice of moving resources from a project in the short term can be effective to ensure the more important or business-critical project finishes on time and below budget. However, Yaghootkar and Gil (2012) argued that increasing the size of a project team to keep the project from falling behind schedule can decrease productivity on other projects because the learning curves increase as knowledge resources switch back and forth between projects. Because most project teams are temporary, they may not have established knowledge management processes and a culture that supports the creation, sharing, and transferring of knowledge among team members. As a result, critical knowledge can be lost after the project has been completed and the team members have moved on to other projects, unavoidably hurting the growth of organizational knowledge and impairing organizational learning (Ajmal et al., 2010).

#### **Problem Statement**

Knowledge loss is one of the most important risk factors facing PBOs today, impacting their project productivity and overall competitive advantage (Martins & Meyer, 2012). PBOs face the risk of knowledge loss through either employee turnover or project termination when the project team is disbanded (Bourouni et al., 2014). Bessick and Naicker (2013) added that an important factor for organizations to remain competitive is the retention of the knowledge that exists within their project team members. Management uses many techniques to manage knowledge, such as knowledge sharing and knowledge transfer; however, it is unclear whether they do enough to prevent knowledge loss. It is up to management to use an employee's knowledge; however, there is always the risk that the employee will leave, taking their knowledge with them (Parboteeah et al., 2016). The general business problem is that knowledge loss impacts project quality, customer satisfaction, and organizational efficiency. The specific business problem is that management does not clearly communicate the knowledge retention strategies needed to prevent knowledge loss that could adversely impact project productivity.

# **Purpose of the Study**

The purpose of this qualitative case study was to explore knowledge retention strategies that help to prevent knowledge loss in PBOs. This study contributed to the existing body of knowledge on knowledge governance, knowledge retention, project knowledge management, and organizational learning. This study included semi-

structured interviews of managers and consultants to determine the knowledge retention strategies managers use to mitigate the risk of knowledge loss. The results from this study could help managers develop new strategies to retain knowledge, which may strengthen an organization's competitive advantage with its customers.

# **Research Questions**

For this research study, I evaluated information from semi-structured interviews with managers and consultants about the impact of knowledge loss on project productivity in PBOs and how to prevent it. The information from these sources addressed the following question:

RQ1. What knowledge retention strategies do managers use to prevent knowledge loss?

# **Conceptual Framework**

I used two concepts to construct the framework for this study: (a) the knowledge-based view (KBV) of the firm (Grant, 1996) and (b) intellectual capital (IC) (Edvinsson & Malone, 1997). Each of these concepts supported assumptions that provide a solid basis for understanding the importance of knowledge in organizations and the strategies management implements to identify, capture, and retain that knowledge.

The KBV of the firm states that knowledge is the most critical resource for an organization (Grant, 1996). Additionally, this view proposes that knowledge increases financial performance for an organization (Mousavizadeh et al., 2015). Knowledge can be created and combined from multiple entities within the organization, such as the

culture, employees, systems, and policies (Tongo, 2013). This view also acknowledges that intangible resources such as knowledge, skills, and talent may contribute more to an organization reaching and maintaining a high performance than tangible resources such as physical and financial assets (Wang et al., 2014). The KBV extends from the resource-based view of the firm, which suggests that rare and valuable resources may give a competitive advantage to the organization that acquires the resources (Brown, 2014). The resource-based view stresses that the correct management of a firm's resources is vital to the success of the firm and is a critical indicator of its competitive advantage (Mousavizadeh et al., 2015). Munoz et al. (2015) stated that the resource-based view arose from the notion that the source of competitive advantage comes from within the organization, and the adoption of new strategies is limited by the level of the organization's resources such as competencies.

IC is defined as the knowledge, skills, and attributes each employee has multiplied by their willingness to work hard (Harris, 2000). Different authors have recognized this explanation of IC as one of the most common definitions of the concept. Secundo et al. (2016) expressed that IC can be put to use to create value for an organization. Demigha (2015) defined IC as organizational knowledge that is used to produce wealth and gain a competitive advantage.

Radenkovic et al. (2014) explained that the concept of IC refers to knowledge as a resource and as a capital. Knowledge as a resource indicates that knowledge is a very important resource that is transformed to create value. Knowledge as a capital considers

knowledge to be an important element that can be validated financially. Gioacasi (2014) explained that IC comprises three different types of capital: human capital, relational capital, and organizational capital. Radenkovic et al. (2014) referred to human capital as the skills and knowledge an individual possesses. Harris (2000) explained that such skills and knowledge increase productivity, which justifies the costs to acquire them. Human capital will be elaborated on further in Chapter 2.

# **Nature of the Study**

The focus of this research study was on the knowledge retention strategies management uses to help in the prevention of knowledge loss in PBOs. I reviewed and considered three different research methodologies for this study. The three research methodologies were qualitative, quantitative, and mixed method. A quantitative research study involves measurement and assumes that the social phenomena under study can be measured. Watson (2015) stated that the measurement is used to analyze data for trends and relationships and to verify the measurements made. The quantitative research methodology was not appropriate for this research study. This study did not involve any measurements and did not set out to analyze data for trends and relationships.

The mixed-methods research methodology was not a proper fit because it requires a purposeful mix of qualitative and quantitative data to draw conclusions based on the total strength of both data sets (Shannon-Baker, 2016). Mixed methods would be appropriate if my study required both qualitative and quantitative data to examine the relationship between variables. I selected the qualitative research approach to gain a

deeper understanding of a workplace phenomenon by gathering data from the participants' point of view. A qualitative research study provided me the advantage of collecting managers, full-time employees, and consultants' perspectives and detailed responses to explain how to reduce knowledge loss to improve project productivity in PBOs.

I analyzed five different qualitative research study designs. I considered grounded theory, narrative, ethnography, phenomenology, and case study. Researchers tend to choose phenomenology when the purpose is to develop a new theory about a phenomenon (Simon, 2013). Simon (2013) stated that researchers use grounded theory to ground a social practice in an environment. Because my research focused on a single unit, grounded theory was not appropriate for this study. The narrative research design focuses on recounting stories to explain life experiences. Narrative research also requires an overview of a person or group through the eyes of the researcher (Jørgensen et al., 2013). I did not select a narrative research design because interviews and artifacts guided my research study.

An ethnography research design would not have provided the required data for my research study because the ethnographer tries to understand the culture of people (Spradley, 2016). Spradley (2016) further argued that ethnography includes techniques, ethnographic theories, and descriptions of human cultures from the perspective of those who have learned them. I did not select the ethnography research design because my study did not address the culture within the PBO. I also did not select a

phenomenological research design because it deals with the meaning of life and the nature of responsibility for personal action and decisions (Van Manen, 2016). I decided to use the case study research design for this study because, as Cronin (2014) explained, case study research uses individual or multiple cases to address a specific situation. The case study design gives the researcher the latitude to study anything in a given situation, whether it is individuals, groups, or a specific phenomenon. Furthermore, the goal of case study research is to create a truthful and thorough description of the case. A more detailed discussion of the research approach and design selection is presented in Chapter 3.

#### **Definitions**

This section includes terms I used that may or may not be familiar to individuals outside of human resources and the information technology (IT) industry.

*Human capital*: the main element of IC (Baron, 2011), comprising knowledge, skills, experience, and professionalism of individuals in an organization (Hadad, 2017; Vidotto et al., 2017).

*Knowledge assets*: any knowledge-based intangible asset or capital that produces value outcomes in the marketplace (Swart & Kinnie, 2013; Wu & Chen, 2014).

Knowledge governance: the adoption of formal and informal organizational structures and mechanisms to have an impact on knowledge management processes and attain superior organizational performance (Cao & Xiang, 2012).

*Knowledge loss*: the reduction of organizational effectiveness and productivity due to the loss of a subject matter expert or knowledge worker (Sumbal et al., 2018). The loss of IC in organizations (Massingham, 2018).

*Knowledge retention*: the capturing of knowledge that is in danger of being lost when an employee leaves an organization (Sirorei & Fombad, 2019).

*Multi-project environment*: an organizational environment where at least two or more projects are executed concurrently while sharing resources from a common pool (Eskerod, 1996).

Organizational forgetting: the unplanned or unintended loss of organizational knowledge (Aydin & Gormus, 2015).

*Project-based organizations*: organizations that produce a majority of products and services through projects for their customers (Chronéer & Backlund, 2015).

# **Assumptions**

There were several assumptions made in establishing the foundation of this research study. The first assumption was that the participants were able to relate their experiences and perspectives about the researched phenomenon. Second, I assumed that management and leadership would provide accurate information regarding knowledge retention practices and strategies. In this study, I assumed that organizations had knowledge management processes to identify, capture, and retain knowledge within the organization. Third, I assumed that the organization had multiple projects varying in size and scope running concurrently. The final assumption was that the projects undertaken

would have most of the risks mitigated, were completed on time and under budget, and met customer satisfaction.

# **Scope and Delimitations**

The scope of this study focused on an information systems (I/S) organization that runs multiple projects concurrently with dedicated teams for each project. Further, the focus was on which knowledge governance mechanisms are in place when project team members are reallocated to another project or leave the organization voluntarily or involuntarily. I delimited the participants of the study to project teams impacted by knowledge loss. The roles of the participants were limited to managers and consultants who work in a PBO.

#### Limitations

The first limitation was the potential for researcher bias in this study. My professional background is in I/S and project management, and I have witnessed how the loss of knowledge can impact project teams and their productivity. Another potential limitation was the sample size of the study. The intended sample size for this research study was 24 subjects. While this size was appropriate for the time frame I selected to study for the impact of knowledge loss on project productivity, it provided a limited view of the case. Because of the sample size, there may not have been complete information on which strategies are used to retain knowledge to improve project productivity. However, the type of semi-structured interview questions and the ability to allow the study participants to give extensive detail and description of their opinions and experiences

provided for a rich and thick depiction of the perceptions that managers and consultants have about their ability to be successful when lost knowledge was not retained by the organization. The level of detail that the sample participants provided helped to balance the small sample size.

The constraints set by the case study design was another limitation of this research study. Purposeful sampling was used for the data collection and analysis of this research study. The main reason for using this sampling technique is that it allows the researcher to identify and select individuals that have knowledge and experience with a specific phenomenon, allowing for the effective use of limited resources (Palinkas et al., 2015). The weakness of using purposeful sampling is that data samples might be biased.

# **Significance of the Study**

Organizations may be served by recognizing the type of impact employee turnover, organizational culture, and leadership can have on their project teams and operations. This study had the potential to fill a gap in knowledge by focusing on preventing knowledge loss through knowledge retention within a PBO. The findings from this study may be used to make project managers aware that failing to manage and capture valuable project knowledge can lead to project delays. The results of this study could provide insights into the processes that an increasing number of organizations use to identify, capture, and retain knowledge at the project level and across the organization.

# **Significance to Business Practice**

Insights into the experiences of managers and employees could help organizations develop strategic approaches to identify, capture, and retain knowledge and IC from employees who (a) voluntarily or involuntarily leave the organization and (b) are reassigned to another business-critical project. Insights based on the themes and concepts of the qualitative information might enable researchers to conduct empirical studies to generalize the findings for future I/S organizations. In addition, this study may influence the operation of organizations in other industries such as banking and finance, education, telecommunications, and health care.

# **Significance to Theory**

This study had the potential to add to the existing literature on knowledge governance, knowledge management, knowledge retention, and organizational learning within a PBO. Pemsel et al. (2014) pointed out that knowledge governance is starting to emerge as an approach to address problems that organizations have with their knowledge management processes. Understanding the application of knowledge governance mechanisms in a PBO may help remove the barriers to knowledge retention and sharing. By not practicing effective knowledge management in PBOs, management is unable to learn from the projects (Akhavan et al., 2014). The failure to document lessons learned from the finished projects can lead to repeating past errors in future projects.

# **Significance to Social Change**

The implications for positive social change may include the continued development of consultants through knowledge retention so they can contribute to the social good of the communities in which they live. Levallet and Chan (2019) noted that knowledge retention is strengthened when strategies such as knowledge transfer and sharing are implemented.

Examining knowledge loss in PBOs may lead to the identification of new information for operational continuity. The new information may assist PBOs to harness and manage the knowledge from their project teams. The findings from this study may contribute to the development of new knowledge retention processes that will benefit project teams and organizations. The information learned from this study could help to retain, preserve, and maintain the project specific knowledge of project team members. The insights based on the themes could help researchers generalize the findings for organizations.

# **Summary**

Knowledge has now become a strategic asset for organizations, and the management of this knowledge is essential because many business activities are knowledge driven. Most businesses organize their work in the form of projects in response to changing business and economic environments as well as increasing competition. In a PBO, projects can vary according to size and scope and, in some instances, run concurrently. Some projects are identified as being critical to

organizational success, and management can decide to move knowledge resources from another project to the critical project. By shifting resources from one project to another, management may fail to recognize the impact knowledge loss will have on the overall productivity of the other projects. Developing a case study from the majority of stakeholders in a PBO based on the capture and retention of knowledge could be a starting point toward addressing that problem.

Chapter 2 includes evidence to support the research problem and research question. Additionally, Chapter 2 is a review of academic literature covering topics of knowledge management, knowledge loss, knowledge retention, and PBOs. The literature review addresses how the organizational factors of employee turnover, culture, and leadership have an impact on knowledge loss. Chapter 3 describes the research approach and design I selected for capturing information about the knowledge retention strategies management uses to prevent knowledge loss in organizations. Chapter 4 describes the results of the collected interview response, including the emerging themes from the study participants and how those themes relate to the literature. Chapter 5 includes the summary and interpretation of the collected data, recommendations, and what those recommendations mean for both theory and practice and the implications for future research.

# Chapter 2: Literature Review

Knowledge management serves an essential role during a project life cycle for two reasons. First, transferring and sharing knowledge positively affects project performance in terms of cost, schedule, and quality. Second, new knowledge needs to be generated and integrated into projects and organizational practice. If the new knowledge is not integrated, the knowledge will become lost or worthless (Oun et al., 2016). Lin et al. (2016) contended that knowledge management helps organizations share insights, increase work efficiency, document experiences from projects, and create and retain IC. When employees leave a project or organization, valuable knowledge is lost if specific knowledge retention mechanisms are not applied. In light of this result, knowledge retention should be an element in an organization's knowledge management strategy (Lin et al., 2016).

The purpose of this research study was to explore knowledge retention strategies to prevent knowledge loss in PBOs. An examination and understanding of the concepts and theories related to managing and preventing the loss of knowledge in PBOs is needed. The goal of this chapter is to investigate these concepts and theories through a literature review. The literature review of this research study begins with the exploration of knowledge and the concepts of knowledge management, creation, sharing, and transfer. Other topics reviewed are knowledge governance, organizational learning, absorptive capacity, and PBOs. The literature on human capital, social capital, and relational capital establishes the foundation for the concept of knowledge assets. In

addition, the literature on employee turnover, organizational culture, and leadership was reviewed to discover how these challenges impact the management of knowledge loss.

# **Literature Search Strategy**

The peer-reviewed articles in this literature review were identified using the Walden University online library, Google Scholar, and the following academic and industry databases: Business Source Complete, ABI/INFORM Collection, Emerald Insight, SAGE Journal, EBSCO, ProQuest Central, Academic Search Complete, ACM Digital Library, ABI/INFORM Complete, Computer and Applied Sciences Complete, ScienceDirect, and IEEE Xplore Digital Library.

During the initial literature search, I used the peer reviewed search option to obtain scholarly articles identified by keywords. When searching electronic databases such as ProQuest, I selected the option *peer reviewed* to obtain scholarly articles, and I limited the publication date to a 5-year time range. Articles published outside of the 5-year window were considered if the initial search failed to retrieve extensive resources on a specific subject. The resulting research criteria built upon and complemented each other, resulting in more than 130 peer-reviewed articles and books. Although some of the older references provided the contextual framework for the study, more than 80% of the overall references were peer-reviewed and published within 5 years of my anticipated graduation date.

I also limited the search terms to title, author, or keyword to refine subsequent searches. The following keywords were used to search the electronic databases: *employee* 

turnover, human capital, intellectual capital, knowledge assets, knowledge loss, knowledge based view of the firm, knowledge governance, knowledge management, knowledge retention, knowledge strategy, knowledge transfer, multi-project environment, organizational culture, organizational performance, organizational learning, organizational knowledge, absorptive capacity, project governance, project leadership, project organizations, project-based organizations, project teams, and social capital. In addition to the journal articles, books on knowledge management were reviewed for definitions, terms, theories, and concepts.

# **Conceptual Framework**

The framework for this study consisted of the theory of the KBV of the firm and the concept of IC. The KBV theory and IC were used as a guide to explore knowledge resources and knowledge management processes and how they impact organizational and project knowledge. The KBV of the firm addresses the point that knowledge is an important factor to an organization and its competitive advantage (Akanbi, 2016; Handzic et al., 2016). The concept of IC refers to the skills and talents of individuals or knowledge-based resources that create value and contribute to the business and economic success of an organization (Hussinki et al., 2017; Kianto et al., 2014).

# **Knowledge-Based View of the Firm**

The KBV of the firm theory is built on the resource-based view of the firm originally proposed by Edith Penrose in 1959. The resource-based view of the firm assumes resources and capabilities are not uniform across competing organizations

(Killen et al., 2012), and these resources are drivers of competitive advantage (Jugdev & Mathur, 2012). Lopez and Esteves (2013) indicated that the KBV theory proposed that knowledge is the most valuable resource to organizations because it is essential to many activities and processes. Erden et al. (2014) explained that how an organization creates, transfers, and uses knowledge has a significant impact on its overall performance and ability to be competitive within a specific industry. Individuals within an organization are the main creators and sources of knowledge. The KBV theory proposes that if knowledge is a leading resource within an organization, and individuals create and store knowledge, employees are the key stakeholders in an organization (Agostini & Nosella, 2017). The KBV theory also enhances the awareness of different types of organizational knowledge such as tacit and explicit organizational knowledge (Mainga, 2017). It is necessary for management to incorporate individual knowledge into the organization by providing an environment where the knowledge can be shared among the specialized knowledge workers. Additionally, the KBV theory strengthens the position that knowledge management processes such as the creation, acquisition, and transfer of knowledge lead to organizational performance (Barkat & Beh, 2018).

# **Intellectual Capital**

Wang et al. (2016) identified another strategic resource for organizations in a knowledge-based economy. That resource is IC. IC results from the knowledge, talent, and skills of each employee (Sayadi et al., 2013). An organization's IC is the accumulation of all its knowledge resources, which exist within or outside the

organization (Hsu & Sabherwal, 2012). There are varying definitions of IC; however, they all seem to agree on the same points. IC is a combination of intangible assets used to generate value in the organization (Tarride & Osorio-Vega, 2013). Fragouli (2015) defined IC as a group of organizational knowledge assets that contribute to the competitive position of an organization by adding value to distinct stakeholders.

Salmaninezhad and Daneshvar (2012) argued that IC consists of different components that are grounded in employees, routines, and customer relationships. Vale et al. (2016) conceptualized IC by breaking it down into three dimensions: human capital, structural capital, and relational capital. Human capital is the knowledge, attributes, attitudes, and abilities stored in employees (Asiaei & Jusoh, 2015; Vale et al., 2016). Structural capital is the organization in general, including systems, culture, and intellectual property such as patents or trademarks (Ienciu & Matis, 2011; Vale et al., 2016). Relational capital is the relationships an organization has with the individuals or organizations to which it sells products or services. These relationships may involve customers, suppliers, and vendors (Dumay & Garanina, 2013; Vale et al., 2016).

Hsu and Sabherwal (2012) emphasized that the literature on IC shares the same broad objective as knowledge management, which is to understand the role of knowledge and its management in an organization achieving success and competitiveness. The literature on IC also looks at the nature of organizational knowledge, its different types, and how it affects overall organizational performance. Hussinki et al. (2017) examined the connection between IC and knowledge management practices and their impact on

firm performance, concluding that firms with strong IC and knowledge management practices outperformed firms with low IC and knowledge management practices. Wang et al. (2016) noted that firm performance is influenced by the relationship between the firm's IC and the knowledge management strategy they choose to implement. Zaei and Kapil (2016) identified that IC has a positive relationship with knowledge management. This relationship helps promote the success of knowledge management initiatives within the organization. The authors concluded that knowledge could be managed effectively by increasing and improving the components of IC. Within a project management context, Cole (2017) referred to project management as an IC commodity with its foundation in knowledge and experience. Turner et al. (2015) used an IC perspective to argue that managing projects uses human, social, and organizational capital, which are the three dimensions of IC. The authors concluded that a combination of human, social, and organizational capital was present in the projects but not all three at the same time. Handzic et al. (2016) added that IC has positive results on the success of IT projects.

# What Is Knowledge?

Before discussing the subject areas of knowledge management and knowledge governance, it is important to explore the concept of knowledge and the different types of knowledge in relation to knowledge management. Many organizations are now becoming interested in the nature of knowledge, mainly in part as a result of IT, which provides the ability to manage knowledge as a corporate asset (Almudallal et al., 2016). Koskinen (2013) stated that the knowledge in knowledge-intensive companies is fundamentally

social in nature because the work done in these types of organizations is intellectual in nature where qualified individuals with specialized knowledge form the majority of the work force.

Brajer-Marczak (2016) reviewed literature on knowledge management in the improvement of business processes and discovered that authors used various definitions of knowledge. Knowledge has been studied from many perspectives and can be problematic in its definition and scope (Randeree, 2006). One perspective states that knowledge comes from structured and organized information resulting from cognitive processing and validation. This perspective of knowledge answers *how* questions (Cooper, 2016). Tang et al. (2016) defined knowledge as the capacity to act effectively where capacity is a trait associated with a person or group.

The significance of knowledge management has encouraged management studies scholars to dig deep into the meaning of knowledge. They have wrestled over the question: What is knowledge? This question has occupied the minds of philosophers for centuries (Almudallal et al., 2016). This question is explored within the philosophical discipline known as the theory of knowledge or critique of knowledge, among other names. This discipline takes the viewpoint that human knowledge is scientific (Segundo, 2002). This approach branches into two main epistemological camps: the objectivist perspective and the practice-based perspective.

The objectivist perspective regards knowledge as an entity or commodity that can exist independently of people in a codifiable form (Hislop, 2013). This perspective also

addresses knowledge as something that exists in the human head and is acquired, modeled, and expressed precisely in the most objective and explicit terms possible (Almudallal et al., 2016). The practice-based perspective contrasts with the objectivist perspective by challenging that conceptualization of knowledge and instead assumes knowledge is embedded in the context in which it arises. As such, it is inseparable from peoples' work, places, and practices (Hislop, 2013). Thus, in this perspective, knowledge is not viewed as an objective entity that can be separated from people or activities (Almudallal et al., 2016).

There are two types of knowledge, tacit and explicit knowledge, which are equally important and complementary to each other (Millar et al., 2016). Explicit knowledge is formal and documented, while tacit knowledge is informal and lives in people's minds as skills and experiences (Ayub et al., 2018; Polanyi, 1966). A further discussion on tacit and explicit knowledge is detailed below.

# **Tacit Knowledge**

Tacit knowledge consists of two elements, cognitive and technical. The cognitive element is an individual's mental model such as their beliefs and viewpoints. The technical component consists of concrete experience, crafts, and skills applied to a specific context (Alavi & Leidner, 2001). Tacit knowledge, also known as implicit knowledge, is currently present in the individual but cannot be expressed easily by the individual in either spoken or written form (Turner et al., 2012). Jasimuddin and Zhang (2014) reviewed the work of several academics who argued that tacit knowledge is

personified in the human mind and cannot be detached from the person or people who possess it, thus making it challenging to retrieve and transfer.

# **Explicit Knowledge**

Explicit knowledge is tacit knowledge that is codified, documented, and shared, making it readily available to either a single person or group at a minimal cost (Jasimuddin & Zhang, 2014). Some examples of explicit knowledge are policy manuals, technical documentation, and reference guides (Terzieva & Morabito, 2016). Explicit knowledge is tougher to comprehend than tacit knowledge because it must be tacitly understood and put into practice (Addis, 2016). Addis explained that all knowledge is established initially as tacit knowledge. This explanation is rooted in the epistemology of practice-based knowledge and implies that explicit knowledge does not exist at first.

# **Knowledge Management**

Knowledge management leverages the KBV of the firm to understand how to use knowledge to achieve organizational effectiveness and efficiency (De Toni et al., 2017). It is a discipline that helps to design strategies to make sure that knowledge flows to the correct people when needed (Kianto et al., 2019). The concept of knowledge management has been defined in many different ways by many authors, and none of these definitions are completely accurate nor are they completely imprecise (Terzieva & Morabito, 2016). Hislop (2013) analyzed the literature on knowledge management and emphasized that there is a lack of consensus regarding how knowledge management is defined and conceptualized. Knowledge management is a process of capturing

knowledge from either individuals or groups and sharing it with the desired entities. This process makes sure that knowledge reaches the right people so they can make the right decision and raise organizational performance (Addis, 2016; Gitinejad & Keramati, 2013). From a pragmatist perspective where knowledge is defined as the outcome of inquiry, reflective knowledge management is another approach to knowledge management. This approach fights for the participation of all knowledge workers in the sharing of knowledge with others (Vo, 2012).

Knowledge management is a concept that crosses over different disciplines. It can be applied to business, I/S, and organizational management. Reich et al. (2012) indicated that knowledge management helps to create definite bodies of knowledge within a project. The knowledge created is important to the achievement of the project goals. Even though some of the knowledge will stay tacit, most of the knowledge needs to be made explicit so it can be reviewed, confirmed, shared, and finalized. Reich et al. (2014) developed a model for project-based knowledge management, which proposes that there are two goals of knowledge management. The first goal is creating an avenue where the business value of the project can be reached. The second goal is making sure there is an understanding between the project team and business sponsors on what the project must accomplish for the business. These goals ensure a plan is in place for the project, which should achieve the desired business results. Additionally, these goals also ensure that there is a shared understanding that project teams recognize consequences when they deal with unanticipated project changes.

Massingham (2014) provided useful insight about knowledge management as it pertains to managing knowledge resources. This insight known as the product versus process view offers two different perspectives on how knowledge is viewed within the individual and how it can be utilized. The product view of knowledge management attempts to separate knowledge from the individual. This view is about creating explicit knowledge from tacit knowledge and sharing that knowledge with other individuals, teams, or the organization at large. The process view of knowledge management, as Massingham (2014) stated, keeps knowledge with the individual. This view approaches knowledge management as a vehicle to share knowledge through collaborative and cooperative efforts.

# **Knowledge Creation**

Knowledge creation is a collaborative process that focuses on generating new knowledge in terms of new ideas and solutions (Kianto et al., 2019). It is a complex activity that organizations should not perform in isolation. Organizations need to acquire new knowledge from outside sources and combine them with their own internal resources to keep producing knowledge (Zaragoza-Sáez et al., 2016). Tissayakorn et al. (2013) contended that knowledge creation stresses organizational learning and employee motivation for innovation and capturing the lessons learned from previous projects to obtain new and better knowledge.

Knowledge creation is a constant process that takes place through the interactions between individuals and their environment, leading to the development of new

knowledge in the organization (Little & Deokar, 2016). García-Fernández (2015) stated that knowledge creation is understood as the dynamic process of collecting data, converting it into information, and turning it into knowledge, which then progresses through various levels of learning. Nonaka and Takeuchi (1995) created the SECI model which helped in the development of the organization knowledge creation theory. The SECI model describes the interaction between tacit and explicit knowledge through four modes or processes: socialization, externalization, combination, and internalization. Each one of these modes characterizes a different activity in an organization which separately results in the creation of knowledge. The socialization process converts tacit knowledge into new tacit knowledge. The externalization process converts tacit knowledge to explicit knowledge. Combination is the process that creates new explicit knowledge from existing explicit knowledge, and the internalization process creates new tacit knowledge from explicit knowledge.

# **Knowledge Sharing**

Knowledge is an essential element for an organization to survive in today's competitive business environment (Asrar-ul-Haq & Anwar, 2016). Lee et al. (2015) mentioned that knowledge enables organizations to be innovative and improve operational efficiency. However, individual employees prove to be a major hurdle to effective knowledge management. Yen et al. (2015) noted that people should be encouraged and comfortable to share their knowledge in familiar and friendly settings.

The sharing of knowledge is gaining importance in organizations looking to enhance their effectiveness and increase their competitive advantage (Amayah, 2013).

Knowledge sharing is a cornerstone for organizational success and many organizations adopt it as a survival strategy. Knowledge sharing is the transferal of knowledge among individuals, groups, and organizations (Asrar-ul-Haq & Anwar, 2016). Lee et al. (2015b) viewed knowledge sharing as the revealing of pertinent knowledge without transmitting all of it. Knowledge sharing is also seen as receiving feedback from a manager or another employee to help in solving problems and in the development of new ideas or products (Park & Kim, 2015). Knowledge sharing can take place through written or verbal communication and networking with other subject matter experts.

Knowledge sharing occurs at different levels of the organization. Knowledge is shared among employees as they learn new methods of completing assignments in a more efficient and effective way. The knowledge shared at the organizational level involves exchanging knowledge from employees to different groups, where the knowledge is stored and made available for others to use (Yeo & Marquardt, 2015).

It is important to remember that some individuals will not share their knowledge to protect their status in the group. Organizations encourage knowledge sharing by providing incentives, but employees are still hesitant to share. Therefore, it is critical to identify which factors will increase the knowledge sharing behavior of individuals (Lee et al., 2015b).

## **Knowledge Governance**

Knowledge sharing is crucial to reach alignment and a mutual understanding because the "congruence of knowledge cannot readily be achieved without sharing" (Reich et al., 2014, p. 592). Sanz and Ortiz-Marcos (2020) added that knowledge governance has an influence on knowledge sharing by defining how an organization manages the activities related to knowledge. Knowledge governance is a concept that is constantly evolving. The focus of knowledge governance is coordinating knowledge processes that are in alignment with set objectives through governance initiatives (Pemsel et al., 2014). It began from the premise that governance mechanisms can be influential on knowledge processes (i.e., creation, retention, and sharing), which help an organization view knowledge as a strategic resource (Foss, 2007). Karvalics and Dalal (2013) added that the concept of knowledge governance has evolved over the years and, in its latest form, "refers to choosing structures and mechanisms that can influence the processes of sharing and creating knowledge" (p. 3).

Knowledge governance intersects with different disciplines such as knowledge management, human resource management, organization studies, and project management (Foss, 2007). Pemsel et al. (2014) explained that the concept of knowledge governance is closely related to the concepts of knowledge management and organizational learning. Learning is made possible within knowledge governance by producing knowledge and sharing ideas in social networks (Gerritsen et al., 2013). Knowledge governance serves to stimulate purposeful knowledge sharing through

various formal mechanisms, such as organizational structures and reward systems; relational mechanisms, such as steering committees and expert panels; and informal mechanisms, such as trust and organizational culture (Pemsel et al., 2016).

Knowledge governance helps to define a PBO's knowledge-based objectives by providing direction to the knowledge-sharing processes within individual projects and serves to align them with the broader goals of the organization (Ali et al., 2018).

However, Pemsel et al. (2014) stressed that the challenges associated with knowledge governance are especially significant in PBOs, which are temporary by nature.

Knowledge governance mechanisms are necessary to obtain the greatest advantage from the knowledge created through project activities (D'Armagnac, 2015). Ghosh et al. (2012) argued that organizations that do not have a formal knowledge governance in place should adopt a project-focused knowledge governance approach. Pemsel and Müller (2012) focused specifically on knowledge governance mechanisms that would integrate project knowledge at the organizational level and observed that PBOs attempt to implement knowledge governance practices that combine various perspectives.

# **Organizational Learning**

Organizational learning is a core organizational capability that creates competitive advantages (Molodchik & Jardon, 2015). It plays an important role in the acquisition, dissemination, and use of knowledge to adjust to an evolving external environment (Park & Eun-Jee, 2018). Cyert and March, who initially proposed the concept of organizational learning in the 1960s, stressed the importance of learning by

experience and the ways in which an organization may adapt to environmental changes (Castaneda et al., 2018). The concept of organizational learning was expanded upon by Argyris and Schon (1978) through the introduction of single and double loop learning as the core element of organizational flexibility.

Organizational learning appears in knowledge management processes at all levels of an organization (Argote, 2013). Odor (2019) viewed organizational learning as a means of gathering information to create knowledge and using that knowledge to improve the organization. An organization's knowledge base is constantly changing when the organization acquires new experience. This perspective stresses the role that experience has on the process where knowledge is accrued in the organization as its employees either perform or try to perform tasks (Echajari & Thomas, 2015). Organizational learning is the capacity of an organization to acquire the knowledge necessary to survive, sustain, and compete in its environment (Sathishkumar & Karthikeyan, 2017). Brandi and Iannone (2015) identified three perspectives on organizational learning. The first is learning by encoding inferences from history into routines that guide behavior, the second is a process of detecting and correcting errors, and the third is a constant evolution through social and cultural interactions. The three perspectives show a shift from formal learning to informal learning and are applicable in various organizational structures from bureaucratic (formal learning) to cultural structures (informal learning).

Organizational learning and knowledge management are two distinct concepts that in specific ways complement each other. Some of the significant components of organizational learning are knowledge creation, acquisition, sharing, and application. These components correspond to most definitions of knowledge management processes (Castaneda et al., 2018; Odor, 2020, Kordab et al., 2020). Hammoud (2020) summarized that organizational learning focuses mainly on the processes by which knowledge is created, acquired, stored, shared, and applied. One of the most prominent organizational learning features is the organization's ability to recognize the need to change and adjust. Knowledge management aims to create the organization's values through knowledge management processes (the creation, capture, storage, transfer, and implementation of knowledge). These values emphasize the need for individual knowledge to be available for everyone in the organization (Mitrevski & Aceski, 2017). Jaber and Caglar (2017) also asserted that organizational learning is complementary to knowledge management. Organizational learning is crucial in entrenching the knowledge gained into the fabric of a particular organization. However, it is entirely dependent on individual learning within a particular organization, while knowledge management is more of sharing within the institution.

Qi and Chau (2018) explained that knowledge management is closely related to organizational learning because it is a critical capability that provides organizations with a source of competitive advantage. Sathishkumar and Karthikeyan (2017) contended that organizational learning is accelerated through knowledge management by creating a

common knowledge repository, identifying and codifying competencies and routines, and manipulating information within and external to the organization. Castaneda et al. (2018) conducted a review of organizational learning and knowledge management research to determine if organizational learning was conceptually absorbed by knowledge management. They summarized that knowledge creation, acquisition, and transfer are characteristic processes of organizational learning. Another key element of organizational learning and a firm's competitive advantage is absorptive capacity.

## **Absorptive Capacity**

Absorptive capacity is very important in maintaining the existence of the organization as it relies increasingly on external knowledge for enhancing innovation and performance (Rezaei-Zadeh & Darwish, 2016; Supartha & Kumala Ratih, 2017).

Absorptive capacity is the ability of an organization to discover external knowledge in its environment. Once the external knowledge is discovered, it is acquired and incorporated into its knowledge management processes (Aribi & Dupouet, 2016). Cohen and Levinthal (1990) initially presented the concept of absorptive capacity as the ability of an organization to recognize the value of new external information and assimilate and apply it to existing projects. Absorptive capacity is influenced by prior knowledge, which regards basic skills and enables an organization to recognize, assimilate, and apply new information, contributing to innovation and organizational performance (Mariano & Walter, 2015). Grandinetti (2016) added that the constitutive elementary processes of absorptive capacity are the monitoring and assessment of new knowledge and its

assimilation and the subsequent use of this newly integrated knowledge. Mariano and Walter (2015) linked absorptive capacity to knowledge management processes such as acquisition, creation, and sharing.

## **Project-Based Organizations**

More organizations now use projects, programs, and portfolios as primary methods for delivering new products and services in the current global economy. This new delivery method causes companies to organize in a project-based structure (Schacht et al., 2015). The PBO is different from other organizational structures. A PBO may either be a stand-alone company making products for external customers or a subsidiary of larger firms producing for internal or external customers (Koskinen, 2010). Stulgiene and Ciutiene (2012) added that PBOs are split into two groups: One group of PBOs carries out management by projects, and the second group gets their revenue through subcontracting with other companies under contract.

The PBO develops a majority of its products based on custom designs for its customers (Akhavan et al., 2014). In a PBO, the project is the main mechanism for coordinating all the key business functions of the organization (Rajhans, 2018). Projects are known as foundations of innovation and permit functions that cross the periphery of the organization. The PBO gets a benefit from the distinct and innovative nature of projects because project team members have the ability to be more creative and let fresh, new ideas emerge (Moud & Abbasnejad, 2012). An advantage of the PBO is that its

flexibility makes it well suited to incorporate different streams of knowledge and take on complex tasks (Miterev et al., 2017).

There are numerous benefits associated with the adoption of a project-based organizational structure. Some of the benefits are higher output quality, the ability to respond quickly and flexibly to each customer's needs, and the ability to innovate in collaboration with clients and suppliers (Di Vincenzo & Mascia, 2012). Bourouni et al. (2014) emphasized that PBOs have a higher level of flexibility than other organizational forms when applying and integrating different types of organizational knowledge and skills.

Di Vincenzo and Mascia (2012) contended that PBOs also present considerable challenges in promoting organization-wide and project-to-project learning. Projects function as distinct entities where the project team members are often geographically dispersed, potentially causing communication gaps among team members and between projects. Another consequence for geographically dispersed projects and teams is the knowledge transfer and learning process is hampered. This impacts effective project communication and learning from other projects (Akhavan et al., 2014). Projects have a tendency to isolate their team members from other professionals and their peers, leading them to become knowledge silos. Since projects are temporary, the lessons learned from the project after it ends and the teams are dispersed to other projects are sometimes not documented (Bashouri & Duncan, 2014). Based on these difficulties, Handzic et al. (2016) identified two challenges facing PBOs. The first is recognizing the current

project-related knowledge resources and what additional knowledge resources are needed to improve the rate of project success. The second challenge is understanding what mechanisms are needed to better manage these resources.

## **Knowledge Assets**

Knowledge assets are beginning to become a crucial aspect for an organization's overall performance and competitiveness (Gomezelj & Antoncic, 2015). Knowledge assets are a critically intangible resource and not recorded on the balance sheet; however, they do have an impact on an organization's bottom line and competitive advantage (Killingsworth et al., 2016; O'Donoghue & Croasdell, 2009). Knowledge assets are seen as the accumulation of organizational skills and know-how in individuals that make the organization more competitive in the marketplace (Chun & Yoong, 2015). Swart and Kinnie (2013) viewed knowledge assets as a knowledge-based capital that creates value-based outcomes in the marketplace.

Knowledge assets represent various forms of capital. Mura et al. (2016) contended that knowledge assets are represented by organizational capital (knowledge available to employees via explicit knowledge) and social capital (tacit knowledge retrieved through collaborations with coworkers or clients). O'Donoghue and Croasdell (2009) viewed it in terms of human, structural, and innovation capital. Each of these knowledge assets can be joined to create IC, which takes the form of goods or services developed for customers (Swart & Kinnie, 2013).

Organizations may need to use different types and combinations of knowledge assets. It is important that organizations understand how knowledge assets are utilized in different business units and in an employee's role and which type of knowledge assets are important to their success (Handzic et al., 2016). Smith (2003) argued that a problem with identifying an employee's knowledge and abilities as an asset of the organization is that the organization can't possess people or their talents. The organizational assets that provide information and services reside in the human and social capital within that organization. Therefore, how well an organization can leverage their knowledge assets to create value can have a significant impact on its performance in the long term (Killingsworth et al., 2016).

# **Human Capital**

People are a vital resource for all organizations, as they contribute to organizations' survival, development, and competitive success (Dădârlat & Dumitrașcu, 2015). Within an organization, employees are the most important source of competitive advantage and, consequently, of business performance (Gomezelj & Antoncic, 2015). The combined intelligence, skills, and talent within an individual gives an organization its unique character and is viewed as human capital (Massingham & Tam, 2015). Human capital has a positive influence on an organization's ability to be innovative and strategic (Vidotto et al., 2017).

Human capital refers to the skills, knowledge, talent, and experience that lives in the individual as well as strategic competencies that can't be easily imitated or copied (Nafukho, 2009; O'Donoghue & Croasdell, 2009). The previous definition of human capital is not limited to just knowledge and skills. It also includes competencies that should be put into practice to develop organizational activities. Human capital is rooted in the members of an organization. Wang et al. (2014) explained that human capital represents the individual knowledge stock of an organization to achieve specific goals. The human capital theory proposes that people possess skills, knowledge, and abilities that provide economic value to organizations (Seleim & Khalil, 2011). Nafukho (2009) added to the theory by stating that individuals are motivated to invest in themselves in different ways by buying education and training. Vidotto et al. (2017) noted that the human capital perspective identified people as an asset that needs to be cultivated. Investments should be made in workplace training, health, and economic information. Furthermore, the investments should cover knowledge, skill, talent, behavior, commitment, and time. When employees are trained, their earnings will increase based on productivity (Nafukho, 2009).

# **Social Capital**

Individuals rely on vibrant, creative, and trusting relationships (social capital) to produce valuable outputs (Swart & Kinnie, 2013). Social capital creates communication channels to bring about the creation and sharing of knowledge between individuals, groups, and colleagues (Bharati et al., 2015). Ramadan et al. (2017) argued that social capital provides the method for exchanging knowledge back and forth between social network members. Social capital is about social relationships with family, friends, and

colleagues (Felicio et al., 2014). Such social relationships give admittance to prized resources such as information, influence, and camaraderie, which empower action. Furthermore, social capital helps to facilitate the exchange of knowledge resources between organizational units, including business units, and projects (Bartsch et al., 2013). Furthermore, social capital focuses on creating strong ties between the organization and its clients (Nafukho, 2009). Social capital has varying definitions from different authors on the topic. Lee et al. (2015a) defined social capital as an integrated idea of resources, whether actual or potential, that an individual or a group acquires through a social system. Choi (2015) viewed social capital as a resource rooted in the relationships of individuals, communities, and networks.

Yu et al. (2013) contended that social capital provides stimulation for individuals to participate in sharing knowledge within work teams. Social capital has a valuable role in inspiring work groups to share knowledge because knowledge sharing is recognized as a collaborative activity that creates a major benefit for the group (Choi, 2015). A special form of social capital that is present in the relationships between team members is team social capital (Lee et al., 2015a). Team social capital can also be characterized as team bonding. It serves the purpose of a social glue, tying all team members together to work for a shared goal (Han, 2018). Lee et al. (2015a) related the building of team social capital to the level of knowledge on an I/S development project team as well as knowledge sharing. Lee et al. added that team social capital built and sustained among

team members serves as the starting point for knowledge sharing and working collaboratively to solve problems on I/S development projects.

Social capital serves as a catalyst for implementing knowledge management and a broker in the knowledge creation and IC relationship (Seleim & Khalil, 2011). Bharati et al. (2015) added that social capital enables knowledge management activities within and across the organization. The management of knowledge involves social interactions throughout the organization, and these interactions focus on the identification, creation, and sharing of knowledge between individuals, groups, and organizational units. At their core, these knowledge processes are supposed to assist in the building of social capital by building communities of practice (Ramadan et al., 2017).

# **Knowledge Retention**

Knowledge loss has become a significant issue that could make organizations vulnerable in difficult economic times as well as during economic growth periods when the competition is widespread (Martins & Meyer, 2012). Since organizations cannot afford to lose knowledge, they need to retain knowledge from their employees before the employees leave. Knowledge retention is a process that organizations can use to reduce the risk of knowledge loss through processes like mentoring and coaching (Bratianu, 2018). Bairi et al. (2011) described knowledge retention as an action-oriented, grounded method of addressing the threat of knowledge loss.

Sumbal et al. (2018) suggested that to retain knowledge in an organization, there are three steps in the process: (1) identifying critical knowledge, (2) transfer critical and

undocumented knowledge, and (3) integrate the retained knowledge for reuse in business processes. In a multicase study of four organizations, Daghfous et al. (2013) concluded that strategies that focus on the retention of knowledge and its integration into the organization's processes and routines would be useful to mitigate knowledge loss.

Various strategies such as training, social networks, communities of practice, succession planning, and leveraging retired knowledge workers can be used to retain knowledge in organizations (Chigada & Ngulube, 2016; Makhubela & Ngoepe, 2018).

Knowledge retention can have both positive and negative effects on an organization. Bessick and Naicker (2013) identified talent management (recruiting and mentoring), job satisfaction, and organizational commitment as barriers to knowledge retention. Schmitt et al. (2011) argued that retained knowledge hides inefficiency and rigidity and can prevent adjustment to new situations. On the positive side, knowledge retention helps to control transactional costs. A challenge facing organizations is the lack of a defined knowledge retention process and understanding of its importance (Raudeliūnienė et al., 2018). Wikström et al. (2018) contended that knowledge retention needs to be well integrated into an organization's business operations and should start before key employees are about to leave or retire. Makhubela and Ngoepe (2018) recommended that organizations enforce knowledge retention policies in order to achieve their objectives.

## **Knowledge Loss**

The key to success for organizations in the current economy is to leverage their ability to use existing knowledge to create new knowledge. Most of the time, organizations ignore the importance of knowledge and fail to capitalize on its benefits. Managing knowledge is an essential rule for organizations to remember (Sumbal et al., 2018). The possible consequence of not managing that knowledge is to lose it without retaining it.

Knowledge loss is the result of a more mobile workforce, employees nearing retirement, employee turnover, and disability (Bratianu, 2018; Massingham, 2018). The impact of knowledge loss can be felt at the organizational level in terms of skills shortages. Rashid et al. (2019) noted that knowledge loss not only impacts project quality and employee productivity but threatens project sustainability. Knowledge loss impacts organizations in different ways. The first impact is the organization's credibility with its customers. The second impact is the length of time to train new employees to replace retiring employees. The third impact is the decrease in revenue, and the final impact of knowledge loss is for organizations to acknowledge the need to establish a knowledge management system.

Knowledge loss is assumed to have a negative impact on an organization; however, Jennex (2014) noted that there are rare instances where knowledge loss can have a positive impact. One such example is when the expertise lost is no longer incorporated into an organizational product or service. Massingham (2008) studied the

impact of knowledge loss through the remaining employees or survivors. These employees are affected by the capital type (human, social, structural, and relational) and activity, such as using social capital to create new knowledge, perform various team or organizational activities, and solve project problems.

The risk of knowledge loss is defined as the potential impact on an organization concerning efficiency and productivity due to the loss of a subject matter expert or knowledge worker (Sumbal et al., 2018). Knowledge loss is gradually becoming an organizational risk for two reasons: The first reason is the changing global demographics that can impact the workforce, and the second reason is employee turnover increasing because of changes in the relationship between employers and employees (Massingham, 2018). The risk of knowledge loss increases when dealing with employees with specialized knowledge and expertise of the organization's goals and strategies. When employees choose to leave the organization, the time frame to transfer knowledge may be shortened, and the shortened time frame creates specific challenges if the employees had specialized knowledge (Levallet & Chan, 2019).

# **Organizational Factors That Can Impact Knowledge Loss**

This section identifies and addresses three factors organizations deal with that can have an impact on knowledge loss. It begins with a discussion of employee turnover, then organizational culture, and concludes with leadership. Each section also includes a brief discussion of how the issue has an impact on knowledge loss, knowledge management, and PBOs.

# **Employee Turnover**

One of the most detrimental problems facing organizations today is employee turnover (Hana & Lucie, 2011). Kanade et al. (2015) stressed that employee turnover is a growing problem for any organization because it can create negative bottom-line impacts. The financial costs of employee turnover can amount to thousands of dollars on an annual basis. Furthermore, employee turnover prevents companies from pursuing growth opportunities and acquiring new business. Employee turnover inflicts heavy costs on organizations, both directly in terms of recruitment and induction costs and indirectly in terms of organizational knowledge and skills (Arora, 2015). Devi and Krishna (2016) stated that employee turnover reduces the effectiveness of the organization by impacting its productivity rate and also diminishing the morale of present employees working in the organization. Losing skillful, competent employees creates a problem in teamwork and leads to the incurrence of extra costs for replacements (Kuruppuge & Gregar, 2018). The financial and organizational costs caused by employee turnover can have a lasting impact on an organization.

There are varying definitions of employee turnover. Chauhan and Patel (2013) defined employee turnover as a replacement cycle, in which a new employee has to be hired and trained against a vacancy. Novak et al. (2013) understood employee turnover as the final (permanent) departure of employees from the organization. It occurs when an employee leaves the organization and has to be replaced. Another term considered for employee turnover is attrition, which is defined as a reduction in the number of

employees in an organization through layoffs, resignation, retirement, or death (Arora, 2016; Showry & Sayulu, 2017; Umasankar & Ashok, 2013). Voluntary and involuntary are two separate and distinct categories of employee turnover. Voluntary turnover is started by the employee (Chauhan & Patel, 2013; Marsden, 2016), and involuntary turnover is initiated by the organization to dismiss an employee because of poor performance, organizational restructuring, or the bankruptcy of a business (Chauhan & Patel, 2013; Marsden, 2016). Marsden (2016) added two more categories for employee turnover, as follows: Functional turnover is when an organization fires or loses employees whose talents are easy to replace. Dysfunctional turnover is the exit of employees who are high performers with hard-to-replace skills. Dysfunctional turnover can also erode an organization's workforce diversity when some of the departures are women or members of a minority group.

The impact of employee turnover can have both desirable and undesirable effects. Some of the undesirable effects include the organization's failure to use the knowledge gained by the leaving employee, a decrease in employee morale, and increased demands placed on other employees during the employee turnover period (Hana & Lucie, 2011). Lin et al. (2016) explained that high employee turnover rates can impact employees' performance, their social network at work, and knowledge sharing, while hindering the knowledge flow across organizations. Johansen (2013) also added that high turnover can affect employee and organizational morale and disrupt the organization's social and communication patterns. Employee turnover may benefit the organization in some cases

by having desirable effects (Johansen, 2013). Hana and Lucie (2011) identified the desirable effects of employee turnover, such as new recruits bringing in new contributions and ideas, no stagnation within the organization, and the opportunity for a more suitable employee with broader knowledge and experience to be recruited and hired.

## **Organizational Culture**

Culture is important for companies and other organizations to function well (Ilies & Metz, 2017). It is sometimes used to show the climate, practices, values, and beliefs that organizations develop through their members (Chitsazan et al., 2017). Dauber et al. (2012) explained that organizational culture is seen as a key factor in examining organizations in different contexts such as its importance to establish competitive advantages and its impact on organizational performance. Organizational culture can even be a source of sustainable competitive advantage because it generates a mechanism to adapt to change, gain influence, and stay competitive (Widjaja & Kuslina, 2018).

Organizational culture is the pattern of basic assumptions shared by a group as a result of their experience (Daher, 2016; Widjaja & Kuslina, 2018). Fong and Kwok (2009) viewed organizational culture as the values, assumptions, and expectations that serve as a filter through which strategies are determined. Yazici (2011) described organizational culture as a complex set of knowledge structures that members of the organization use to generate social behavior. Ganescu and Gangone (2017) added that organizational culture is the accumulation of all shared, taken-for-granted assumptions

that a group has learned throughout the organization's history. The term *organizational culture* indicates a typical view of an organization characterized by a stable set of meanings and is a vital factor in influencing behavior and results within organizations (Ilies & Metz, 2017).

Organizational culture is prominent in the construction of IC (Asiaei & Jusoh, 2015). It is pivotal to the value of IC because of the impact it has on the development of key components of IC, specifically human and social capital (Nhon et al., 2018). Nazari et al. (2011) recommended that an organizational culture should not be changed to adapt to an IC management system because it would be more challenging to change the culture since it is deeply entrenched in the organization. Not only has organizational culture been shown to have an integral value to IC, it is also viewed as a component of IC. Sanchez-Canizares et al. (2007) proposed a model to measure IC where culture capital (a combination of national culture and organizational culture) is considered as the main capital. Khoramin et al. (2014) reviewed different models of IC to show that culture is recognized as the main capital. The connection between organizational culture and IC shows that culture is a component of and has value to the development of IC.

## Leadership

Leaders have an important role and significant position of influence within their organizations and on the performance of their team (Micic, 2015; Yang et al., 2014).

Micic (2015) added that the influence of leaders is seen in the effect of their ideas, their ability to inspire change, and their capacity to constantly learn and share knowledge with

other members of the organization. Leadership is a process where an individual influences and motivates a group to achieve a common objective (Banerjee & Ray, 2016; Koohang et al., 2017). Koohang et al. (2017) argued that effective leadership leads to job satisfaction, sound knowledge management, and improved organizational culture and performance. Leadership is an important concept, and effective leadership is required at all levels of an organization.

The role of leadership in managing knowledge is important to organizations (Yang et al., 2014). Leaders deal with knowledge at three different levels: the individual level, the group level, and the organizational level (Riaz & Khalili, 2014). Singh (2008) stressed that the goal of creating and managing knowledge for competitive advantage in organizations is facilitated by the kind of leadership practices that are in place. Micic (2015) conducted a study that identified four leadership styles (charismatic, transformational, team, and network) and the role each style has during the knowledge management process. In that study, Micic concluded that each phase of the knowledge management process needs a different leadership style adapted to it. Koohang et al. (2017) studied that the implementation of knowledge management processes in organizations is promoted by sound leadership. The knowledge management implementation improves organizational performance.

#### **Summary and Conclusions**

Knowledge is seen as a major driver of business success and a powerful asset for organizations (Pemsel et al., 2016; Ren et al., 2018). Knowledge management plays an

important role in organizational productivity, efficiency, and competitive drive (Singh & Gupta, 2014) because it facilitates learning and enhances the capacity to adapt and be flexible (Mageswari et al., 2016). The literature review emphasized two knowledge management processes: knowledge creation and knowledge sharing. The accumulation of knowledge in organizations, also known as knowledge creation, is a combination of internal knowledge development and external knowledge assimilation, otherwise known as absorptive capacity (Martelo-Landroguez & Cepeda-Carrión, 2016). Knowledge sharing is crucial for knowledge management because individual knowledge is not useful to the organization unless it is shared (Mageswari et al., 2016). Mageswari et al. (2016) further noted that knowledge sharing is a crucial knowledge management process for knowledge creation, productivity, innovation, and quality improvement. When knowledge is not shared or retained, it can be lost to the organization.

The gap in the literature identified in relation to this study is how factors such as employee turnover, organizational culture, and leadership increase the risk of knowledge loss. Knowledge retention can help to lessen the risk of knowledge loss; however, what is unclear is the knowledge retention strategies that managers implement to help mitigate the risk. Ajmal and Koskinen (2008) noted that organizational and professional cultures need to be merged together to promote useful knowledge management within PBOs. The impact of employee turnover on an organization results in knowledge loss, which, according to Massingham (2018), can be addressed with appropriate knowledge management. The current research is needed to explore and understand management and

leadership's role in managing knowledge assets and which proactive measures are taken against the issues presented in this chapter. Chapter 3 includes an examination of why I selected a qualitative research approach for this research study and the suitability of the case study design. The chapter also presents the research problem, participant process selection, data collection method, and implications of the research method so that other researchers can replicate this study.

# Chapter 3: Research Method

The purpose of this case study was to explore knowledge retention strategies to prevent knowledge loss in PBOs. The results of this study may help management develop a knowledge governance approach to ensure that existing knowledge management processes are effective in addressing knowledge retention. Chapter 3 includes a description of the research design and rationale as well as the method and the justification for adopting the research approach. The discussion of the method used for this study includes a description of the population of the study, the research design and rationale, and the ethical approaches used to ensure confidentiality and protect the respondents from unwarranted consequences. Also covered in Chapter 3 are discussions on data collection techniques, data collection organization methods, and the data analysis plan. Finally, this chapter addresses issues of trustworthiness with the study.

## **Research Design and Rationale**

The general research question for this study was as follows:

**RQ1:** What strategies do managers use to prevent knowledge loss?

I examined quantitative, qualitative, and mixed-methods research designs to determine the most effective approach to answer the research questions listed above. Cameron and Molina-Azorin (2014) defined the mixed methods approach as a research design with philosophical assumptions that guide the collection and analysis of data and the mixture of qualitative and quantitative data in a single study or series of studies. The central idea of mixed methods research is that the use of quantitative and qualitative

approaches together provides a better understanding of research problems than either approach alone. The research question in this study did not involve statistical measurements and did not need data analysis to identify trends and relationships. I concluded that a mixed method research approach was not appropriate for this research study.

Quantitative research is concerned with collecting and analyzing data that are structured and can be represented numerically (Goertzen, 2017). Zyphur and Pierides (2017) noted that quantitative research is often done in terms of representation and correspondence. This research design can be thought of as a series of *what* questions (Barnham, 2015). However, this research study does not have a series of *what* questions to answer, nor is it concerned with collecting and analyzing data that is structured and can be represented numerically. Based off of this analysis, the quantitative research approach was not suitable for this research study.

I selected a qualitative research approach because the study was about the experiences of participants and the perceptions of leaders. Qualitative research study is mainly naturalistic, interpretive, and inductive (Mayan, 2016). Mayan (2016) further argued that qualitative researchers strive to understand a specific phenomenon and the meaning of the experiences linked to the phenomenon. A qualitative approach has the advantage of providing a deeper understanding of a workplace or social environment through descriptions of the phenomenon from the participants' point of view. Park and Park (2016) asserted that the goal of qualitative research is to explore and understand the

descriptive accounts of various social events, recognizing the similarities and differences between difference accounts of the same event. This method focuses on applied and theoretical findings or discoveries. Florczak (2017) explained that the ultimate purpose of qualitative research is a detailed understanding of a phenomenon. This explanation is in contrast to the goal of quantitative research, which is to calculate and thus control a phenomenon.

Five research design approaches were examined to determine which approach was best suited for this study: grounded theory, ethnography, case study, narrative, and phenomenology (Kruth, 2015; Percy et al., 2015). I did not consider the narrative research approach for this study because it relies on recounting stories to describe personal experiences to elaborate an understanding of life experiences. The stories are tools individuals use to build a sense of their experiences and the vehicles by which they share them with others (Berry, 2016). Ethnographic research was not a suitable approach for this study because it is concerned with describing the culture of people and their behavior (Draper, 2015).

Phenomenology stresses the nature of the human experience and the meaning that people attach to their experiences (Reiter et al., 2011). Phenomenological research is focused on the experience that is being examined, not on the individuals who are having the experience (Kruth, 2015). Matua (2015) added that phenomenological research informs the understanding that reality is best understood from those who have firsthand experience. Mohajan (2018) expressed that the focus of phenomenological research is

relating what participants have in common when they experience a phenomenon. The focus of this research study was not to understand a specific phenomenon through the people who have experienced it, so the phenomenological approach was not appropriate for this study.

Bryant and Charmaz (2012) argued that grounded theory is one of the most popular qualitative research methods utilized across a wide range of disciplines and subject areas. Grounded theory research attempts to create a theory of an event, phenomenon, or experience from an exploration of interviews with participants and coresearchers (Kruth, 2015). However, this study was based on a conceptual framework comprised of the KBV of the firm and IC. Based on the conceptual framework, I did not select grounded theory as a qualitative research approach.

I selected the case study research approach because I intended to look at and understand how to prevent knowledge loss that impacts project productivity in PBOs. Case study research is the exploration of a bounded system from multiple perspectives. The term *bounded system* refers to a single case that can easily be differentiated from other cases. The case is the object of the study, and that case may be an individual, an event, or a series of events clearly bounded and differentiated from other events (Kruth, 2015). Case study research allows the researcher to explore the phenomenon from different viewpoints to get a detailed, balanced image of the phenomenon (Taylor & Thomas-Gregory, 2015).

#### Role of the Researcher

The qualitative researcher plays the role of an instrument through which data are gathered and interpreted. Khan (2014) stated that the researcher should play a neutral role to avoid bias in data collection and to ensure a reasonable interpretation of the results. The case study research approach was an appropriate means for this research study because my role was to gain a deep holistic view of the research problem (Baškarada, 2014). Specific to case study research, Yin (2014) presented the desired attributes a researcher should have: (a) ask good questions, (b) be a good listener, (c) stay adaptive, (d) have a firm grip of the topics being studied, and (e) avoid biases. Being a good listener as a researcher means having the ability to integrate large amounts of new information devoid of bias (Yin, 2014).

My role as a researcher conducting a single case study included (a) designing the study, (b) collecting evidence that related to the research problem, and (c) analyzing and reporting the findings (Yin, 2014). The pool of participants for this study included managers, senior managers, and consultants who work on concurrent projects for a government consulting practice within a consulting firm. Although I work with some of the individuals who formed part of the study sample, none of participants come under my immediate supervision. I am an employee of the consulting firm and have a vested interest in the successful completion of the project. To manage any bias, I acknowledged the expectations I had about the outcome of the research study and avoided

generalizations that aligned with my personal views. My relationships with the study participants are on a professional level.

To prevent any researcher bias from affecting my current work relationships, I used a semi-structured interview protocol. The questions on the semi-structured interview tool were open-ended, with the opportunity for further discussion of each response. The topics that the semi-structured interview tool addressed aligned nicely with the research questions in this study. The semi-structured interviews took place either face-to-face or through a web-based video conferencing service with recording capabilities. I recorded each interview and transcribed the recorded interview after it was completed. One step in the data collection and analysis process involved debriefing the participants after the completion of the interviews and allowing them to review the interview transcript to validate that their responses were what they intended to share regarding the subject matter. At the conclusion of the research study, each of the participants received a description of the findings of the research.

# Methodology

The case study research design was used in this study. The following is a description of the methodology I applied to the case study exploring knowledge retention strategies to prevent knowledge loss in PBOs. The design of this study included semi-structured interviews to collect information on stakeholder perceptions of knowledge loss and knowledge retention strategies within a PBO. I conducted semi-structured interviews using a web-based computer application that could record the audio from the interviews.

This web-based application was free for the study participants and would have allowed me to reconnect with each participant if the audio was interrupted in the middle of the interview. I used a software tool to convert the recorded audio from the interviews into a transcribed document format. Each recorded and transcribed interview went through NVivo software analysis so that I was able to identify trends and themes that were similar across several study participants. The following sections include details regarding the participant selection process, instrumentation, sampling strategy, data collection procedures, and data analysis plan. A plan outlining the data collection and coding process is also discussed.

## **Participation Selection Logic**

Robinson (2014) noted that sampling is an important aspect of qualitative research design; however, it does not receive much attention in research methodology textbooks and journals. Anderson (2010) noted the differences in sampling between quantitative and qualitative research studies. For example, it is important to choose probability samples in quantitative studies so the statistics can be used to give overviews to the population where the sample was pulled. However, in qualitative studies, a smaller sample size is necessary because of the in-depth and thorough work needed.

Purposeful sampling is arguably the most frequently used method of sampling in qualitative research literature today (Gentles et al., 2015). It is a widely used technique for the identification and strategic selection of information-rich cases for the most effective use of limited resources (Palinkas et al., 2015). These cases, by their nature and

substance, will illuminate the question being investigated (Bungay et al., 2016).

Purposeful sampling involves the selection of particular individuals with characteristics relevant to the study who are thought to be the most informative.

Sample size in qualitative research is unclear, because it relies on the answers being sought, the framework, whether that is theoretical or conceptual, the type of data collected, and the amount of resources and time available to the researcher (Butina, 2015). Mason (2010) argued that sample size can vary depending on the methodology used in the research study. Sample size depends on what a researcher wants to know, what the purpose of the research inquiry is, what will have credibility, and what can be accomplished with the available time and resources (Marshall et al., 2013). Smaller sample sizes are used in qualitative research studies because the goal of sampling is to obtain information that is useful for understanding the complexity, depth, variation, or context surrounding a phenomenon (Gentles et al., 2015).

Saturation is a crucial aspect to consider when making sample size decisions in qualitative research (Mason, 2010). Saturation is the point at which the data collection process no longer offers any new or relevant data (Boddy, 2016; Dworkin, 2012). Fusch and Ness (2015) added that a study reaches data saturation when there is enough information to duplicate the study and when additional data coding is no longer viable. However, it is important to note that saturation gives little direction in estimating the sample's actual size before data collection occurs (Boddy, 2016).

Considering the purposeful sampling approach, approximately 24 managers, full-time employees, and consultants who are either currently working or previously worked on a project within a project-based organization comprised the population size.

Participants who are managers were required to have at least 3 years of experience managing a project. Participants who are either full-time employees or consultants should be currently working on a project or previously worked on a project. Knowledge loss directly impacted the population sample size. If some of the potential participants elected not to participate in the research study, there was still a sufficient sample to collect interview data from different perspectives. The tools I used to contact these individuals and to recruit them to participate in the study included the social media platform

LinkedIn and email. I asked the participants if they were willing to respond to semistructured interview questions that reflected their experiences working in a project-based organization and what strategies to retain knowledge were implemented and enforced by management to prevent knowledge loss.

#### Instrumentation

The purpose of this case study was to explore knowledge retention strategies to prevent knowledge loss in PBOs. The data collection instruments used in this study were an interview template and protocol along with an audio conference recording software.

Adams (2010) noted that interviews are one of the main methods of data collection in qualitative research and are used in research that doesn't produce objective or quantifiable data. Hawamdeh and Raigangar (2014) added that an interview is a

communicative event with cultural norms and rules. The interview could be misunderstood or even terminated early if the content or style is inappropriate. Therefore, the interview should be treated as a two-way process of gathering and giving information.

Semi-structured interviews were conducted for this case study. This type of interview was selected because, as Adams (2010) noted, semi-structured interviews are conversations where the outcome is a coproduction of the interviewer and the subject. Adams (2010) also noted that the goal of semi-structured interviews is to explore the experiences of the research participants and the meanings they attribute to those experiences. Rossetto (2014) emphasized that researchers must maintain boundaries to protect the researcher-participant relationship and ethical obligations to do no harm. Roulston (2016) stressed that the successes and failures in generating detailed reports from interviewees are commonly understood as prompted by the interviewer's actions. Efficient management of interactional problems is essential for the success of the interview.

The researcher-developed interview instrument consisted of open-ended questions used to elicit the perspectives of the participants on the strategies to retain knowledge and prevent knowledge loss in PBOs. I used purposeful sampling and semi-structured interviews to address the research questions. Palinkas et al. (2015) stated that purposive sampling allows the researcher to choose the sample unit based on features or characteristics that enhance exploration and understanding of the central themes and questions that the researchers wish to study. The guiding framework of the research

question is a descriptive and interpretive format consistent with qualitative research methodology (Bloomberg & Volpe, 2012). In consideration of this planned approach, the sample comprised managers and consultants with a sample size of 24 personnel from a single unit.

# **Pilot Study**

A pilot study was developed to determine the clarity of the instruments and eliminate potential ambiguity from the questionnaire. Four participants were selected to determine the simplicity of the instructions and interview questions. The participants in the pilot study were disengaged after I adopted feedback. Furthermore, the four participants were selected from the participant pool for the final research study.

Wray et al. (2014) emphasized that a pilot study is a smaller version of the main study used to test whether the instruments of the main study will work as expected. The goal of a pilot study is not to produce results but to check if the sampling strategy is sound, make sure the instruments and data collection schedules are clear, and discover any barriers to the research study (Watson, 2016). The feedback I received from the participants guided me in rephrasing or administering the questions as crafted. The purpose of the pilot study was to determine the level of understanding of the questions by the participants and to remove any ambiguity. The feedback from the pilot study provided additional information that enhanced the quality of the interview questions and the need for additional resources. Responses from the pilot study provided an additional source for confirmability.

### **Data Collection**

This section describes the procedure for data collection, including the location, timing, and individuals involved. This section also includes what characteristics qualified an individual to be a participant in the study. This section concludes with a brief discussion of the data collection protocol, participant privacy concerns, and data saturation.

I collected data consisting of the responses given during semi-structured interviews from managers and consultants who work in a PBO. The boundaries for this case study came from the configuration that all of the participants were currently working on or had previously worked on a project in the preceding 6–12 months under a project management office. All projects under a project management office varied in size and scope. Therefore, a single project management office was the definition of a bounded case. As Yin (2014) noted, obtaining informed consent is an important aspect in conducting ethical research. The letter of consent was presented to each participant for his or her review, acceptance, and signature before applying the instrument. The letter of consent included my contact information, the participant selection criteria, the purpose of the research, and any potential risks resulting from participating in the study. It clarified the voluntary nature of involvement in the research study and provided information on how participants may exit if it became beneficial to withdraw from the study.

The data collection protocol consisted of semi-structured interviews using an audio conference software to obtain the necessary data for the study. The data collection

process was planned to last five weeks. Each interview was recorded. Interview questions were available to the participants prior to the interviews, which lasted approximately 15—30 minutes. If a participant was unable to participate, he or she could respond to the interview questions in writing. All efforts were made to conduct the interview via the audio conference software and only in unusual circumstances were telephone interviews used as another option. The identities of the participants were kept anonymous to protect their privacy. Data saturation was considered to have been reached when no new information was received from the participants.

## **Data Analysis**

This research study was based on the following research question: What strategies do managers use to prevent knowledge loss? The data included information about how participants understand knowledge management processes as well as how effective knowledge management processes are when the organization faces challenges related to employee turnover, leadership, and culture. Whether the participant was a manager or consultant, the interviewer asked all the semi-structured interview questions to obtain information about the role that knowledge management may have played in each individual's perceived effectiveness on their project team.

I took field notes after each semi-structured interview to record my thoughts and impressions of the interview. I used descriptive coding in the first cycle of coding analysis. Descriptive coding allowed me to describe the participants' responses to the interview questions. Descriptive coding is used to summarize in phrases the basic idea of

a passage of data, which contains the content of the message (O'Brien et al., 2018). I employed pattern coding, a process that pulls together several data items into thematically linked categories, and focused coding, which distills the most meaningful themes of the data from the most frequently applied codes, for the second cycle of coding analysis (Miles & Huberman, 1994; Saldana, 2009).

The software for analysis of the collected interview information included UberConference, Dragonfly Naturally Speaking Version 13, NVivo 12 for Windows, and Microsoft Word and Excel for Windows Version 1902. UberConference is a free, webbased application that allows individuals in different locations to conduct an audio conference. If the audio stopped during the middle of the interview, the interview would either have been restarted from the beginning or resumed. The semi-structured interviews were captured via UberConference, resulting in a recording of each semi-structured interview. Dragonfly Naturally Speaking is a tool that processes the recorded interviews from UberConference and converts the recordings into transcripts in rich text format, which I converted into a Word document using the Microsoft Word for Windows software. I believe using Dragonfly Naturally Speaking reduced the need for manually typed transcriptions. If there were any inaccuracies with the transcriptions, I would have listened to the audio recordings while viewing the Dragonfly Naturally Speaking transcripts and made the needed corrections to the transcripts until they matched what I heard the participants say on their recordings. The Microsoft Word transcripts were loaded into the NVivo software in the appropriate format for parsing, coding, and

identification of patterns. I used the NVivo 12 for Windows software to organize the coded information collected from the interviews. This software provided an analysis of any patterns that exist in the collected data. The NVivo package also includes training and an introduction to the most productive methods for using the qualitative data coding software, which are an important element for understanding where unexpected connections may exist within the data after the completion of data collection.

Because of the nature of this single case study, there were likely to be participants whose responses were significantly different from those of other participants or from the expected responses to the interview questions. If a situation were to occur where a respondent's information represented a significant outlier from the other data or did not fit the conceptual framework that underpins the study, I documented the outlier. Patton (2015) noted that data analysis for qualitative research is complex and time-consuming; the challenge is making sense of the massive amounts of data collected. Yin (2014) added that data analysis for case study research is challenging because the methods have not been well defined.

#### **Issues of Trustworthiness**

Connelly (2016) referred to the trustworthiness or rigor of a study as the amount of confidence in data, interpretation, and methods used to ensure the quality of a study.

Using the concept of trustworthiness, Guba and Lincoln (1994) provided alternative criteria for qualitative research, as follows: credibility, transferability, dependability, and

confirmability. This section of chapter 3 will address these alternative criteria for trustworthiness.

# Credibility

Credibility is the believability of the research findings and process. Cope (2014) explained that credibility is improved by the researcher describing their experiences and verifying the research findings with the study participants. Houghton et al. (2013) confirmed that credibility refers to the value and believability of the research. A researcher's experiential knowledge and worldview may introduce the element of bias into the qualitative research process, which may affect the credibility of research findings (Maxwell, 2013). Yin (2014) pointed out that there can be a lack of trust in the credibility of a case study researcher's processes. I ensured adherence to qualitative methods and instruments to achieve credibility for this research study. To maintain validity in qualitative research, member checking is applied (Candela, 2019). Member checking involves asking research participants to review interview data for accuracy; participants may also be invited to review research results (Birt et al., 2016). Baillie (2015) argued that member checking gives the researcher an opportunity to ensure the accuracy of the participants' voice by allowing them to confirm or deny the interpretations of data. Birt et al. (2016) also suggested that the credibility of the research findings is the basis for transferability of those conclusions. I conducted member checking to achieve credibility. Member checking provided an opportunity for the participants to give feedback regarding the interview and any inferences drawn from the comments. An agreement with the

interview transcript through member checking and a confirmation of the absence of bias in my reporting were sufficient to render this study credible. Finally, I ensured clear statements of any bias that may have arose in the course of the study which could have added to the credibility of my research.

# **Transferability**

Transferability of research findings is best described as the criterion for evaluating external validity (Hammarberg et al., 2016). Transferability is defined as whether or not particular research findings can be transferred to a similar study while maintaining meaning from the completed study (Houghton et al., 2013). The nature of transferability, or the extent to which findings are useful to persons in other settings, is different from other aspects of research in that readers actually determine how applicable the findings are to their situations (Connelly, 2016). Yin (2014) pointed out that transferability occurs when a research study offers detailed descriptions of the population of the study, sources of the data collected, demographics, and boundaries of the study. Rigor is used by researchers to create consistent methods to duplicate a study, creating transferability, credibility, dependability, and confirmability of the research (Thomas & Magilvy, 2011). Adopting the rigor presented enhanced the transferability of this study. Additionally, I used thick description as a method of providing external validity for this research study.

## **Dependability**

Dependability is attained when the replication of the study using the same or similar participants and contexts produces the same findings (English, 2015). Munn et al. (2014) added that dependability is recognized if the research process is traceable and clearly documented, while Gelling (2015) explained that dependability demonstrates to readers that the findings are reliable and repeatable. The objective of the case study approach is to ensure another researcher following the same procedures explained by a previous researcher should arrive at the same findings and conclusions (Yin, 2014). Detailed field notes and recordings from the video conferencing were taken from the interviews in the present study. Yin (2014) stressed that each step of the research process should include data on the process itself to ensure reliability.

My research study ensured an audit trail by (a) presenting the purpose of the study, (b) describing the selection process for the study participants, (c) describing the data collection process, (d) demonstrating how the data were interpreted and analyzed, (e) discussing the research results, and (f) communicating techniques to determine the credibility of the data (Thomas & Magilvy, 2011). I checked the interview transcripts to eliminate any ambiguity or mistakes. I also ensured that the coding aligned with the actual meaning by consistently comparing data with the codes. I cross-checked codes by engaging a colleague who helped me to double check the code frequency to ensure agreeability. I used rich, thick description to explain the findings. In-depth description

helps the reader to have a better understanding of the setting and conveys shared perspectives.

## **Confirmability**

Analogous to objectivity in quantitative research, confirmability is the degree to which findings are consistent and could be repeated (Connelly, 2016). The qualitative researcher, as the main research instrument, interacts with the study participants and is responsible for the data analysis. The interviews were transcribed and sent to the participants for validation, ensuring the confirmability of the data in this study and avoiding researcher bias. The transcribed data were stored in NVivo and were to be used, if needed, as an audit trail to ensure that the participants' views were reflected in the data and analysis.

A qualitative researcher who uses reflexivity will be open about their strengths and shortcomings, examines their effect on the research setting, and will note others' reactions to them (Baillie, 2015; Halcomb & Peters, 2016). During this study, I used a reflexive journal to guard against researcher bias and assist in my reflexivity.

#### **Ethical Procedures**

Ethics is a key aspect of social and scientific research. Ethics in research is applicable to the judgment while taking into consideration the actions applied, whether right, wrong, or appropriate, in the entire research processes (Gaus, 2017). Peter (2015) argued that while qualitative research is not as risky as quantitative research, there are

some features of qualitative research that a research ethics committee must be able to recognize to review qualitative protocols effectively.

This research study is subject to review by Walden's Institutional Review Board (IRB) to ensure procedural ethical guidelines were followed. Letters of consent for conducting interviews were provided to the study participants. Copies of each letter were provided to the IRB for the approval process. Participant confidentiality for all information gained during the research was to be maintained.

I sought out participants who were managers and consultants currently working in a multi-project environment. There was no incentive to participate in the study, and participants could withdraw their participation at any stage of the process as stated in the consent form. Information that could identify each participant, project, and organization was omitted to preserve the privacy of the participants and the confidentiality of data. All data collected were to be encrypted as archive files and stored on an external hard drive after the completion of the study and destroyed after three years.

## **Summary**

The purpose of this case study was to explore knowledge retention strategies to prevent knowledge loss in PBOs. The use of semi-structured interview questions served as the frame for discussion of knowledge loss in this study. The interview process allowed for the participants to articulate their experiences with knowledge loss in PBO. Chapter 3 included information on the research design and the rationale for the selection of the qualitative research approach and case study research design. The role of the

researcher and the methodology, which included the recruitment method, choice of participants, data collection, and plan for the coding and analysis of the data, were addressed in this chapter.

### Chapter 4: Results

The purpose of this case study was to explore knowledge retention strategies used by managers to prevent knowledge loss in PBOs. This study included semi-structured interviews of managers and consultants within these organizations. The central question was as follows: What knowledge retention strategies do managers use to prevent knowledge loss? In Chapter 4, I describe the pilot study detailing the impact on the study, the research setting, participant demographics, data collection methods, data analysis procedures, evidence of trustworthiness, study results, and a summary of the chapter.

## **Pilot Study**

I conducted a pilot study with two participants to determine the suitability of the interview questions. The purpose of the pilot study was to determine the level of understanding of the questions by the participants and to identify any potential ambiguity. The pilot study also helped to confirm the suitability of my audio recording instrument. The pilot study enhanced the credibility of the interview questions. The advantage of the pilot study was that it gave me insights on where the research project could fail and where the research protocol could slow down the process.

The feedback I received from the pilot study participants identified ambiguity and redundancy in some of the questions during the interview. The meaning of knowledge retention was understood throughout the interview, and the logic and intention remained consistent. Following simplification of the questions in the interview instrument, the pilot participants agreed that the interview questions were clearer. Confirmability was

enhanced by the pilot study. At this stage, I disengaged the participants from the study. Appendix C contains the interview questionnaires.

## **Research Setting**

There were no organizational conditions that influenced the participants' responses to the interview questions. I recruited the participants in this research study using the LinkedIn networking platform. The selection criteria I used to determine the participant's role was (a) a manager with 3 to 5 years of experience managing a project team, (b) a consultant with 2 to 5 years of experience working on a project team, or (c) a full-time employee with 2 to 5 years of experience working on a project team. Once I identified the potential participants, I followed up with them via email. I sent them a link to the online consent form and a brief survey to determine whether they were a manager, consultant, or full-time employee. Once the participant gave consent, I scheduled an interview on an agreed-upon date and time via email. I conducted the interviews via Zoom with limited to no interruptions. The semi-structured interview format enabled participants to become and remain fully engaged during the interview. The participants were aware of the confidentiality agreement and expressed themselves openly and without incident.

### **Demographics**

There were eight participants in the study, all of whom worked in a PBO. Five of them were managers, and three were full-time employees. All of the participants were at least 30 years of age, and none of the participants were in any way vulnerable as a result

of participating in this study. All of them were able to provide their perspectives and insights within their organizations. Table 1 provides demographic information of the eight participants. All of the interviews were conducted using Zoom, which captured the video and audio recordings. Each of the interviews ranged from 25 to 42 minutes. The given pseudonyms are in XY format so that the X is presented by the letter M for *Manager* or C for *Consultant* and Y is the number identifier assigned to each participant.

**Table 1**Participant Profiles

	Gender and age	Job role	Pseudonym
Participant 1	Male, 30+	Manager	M1
Participant 2	Female, 30+	Full-time employee	C1
Participant 3	Male, 30+	Full-time employee	C2
Participant 4	Male, 30+	Full-time employee	C3
Participant 5	Female, 30+	Manager	M2
Participant 6	Female, 30+	Manager	M3
Participant 7	Female, 30+	Manager	M4
Participant 8	Female, 30+	Manager	M5

#### **Data Collection**

The data collection process began following IRB approval from Walden

University (IRB Approval #04-13-20-0237317). To ensure that the perspectives came

from a balanced group of participants, I attempted to have a proportional number of

managers and non-managers. I employed a purposeful sampling strategy to obtain enough

participants. Using this strategy helped identify participants for each group. There were

some time gaps in the interview activity while I continued to identify and vet a sufficient

number of participants to represent each group for the study. I conducted the interviews

using the semi-structured interview protocol. Data collection began in April 2020 and

continued through September 2020. The extended data collection period was necessary

because of the difficulty in identifying a sufficient number of participants for the study.

The data collection process concluded when the data analysis of interviews uncovered no

new themes, compelling me to infer the presence of saturation.

I collected all interview recordings using the Zoom software, which allowed the participants to communicate via telephone or internet. The Zoom software had an option to record each interview and maintain the recording in an account that was password protected. I informed each participant when I turned on the recording and when I turned off the recording. I used Temi.com to transcribe the interviews. Temi is an online speech-to-text transcription software. Because of the diversity in speech patterns among the study participants, the initial transcription produced by the Temi software had several portions that were not decipherable. As a means of addressing that issue, I did a

secondary review of each interview transcript. The completed interview transcriptions are maintained on a password-protected computer. Because both of the interview protocols contained a specific set of questions, there were no unusual circumstances in the interviews. I posed each question listed in the interview protocol objectively. Due to the interview protocol's semi-structured nature, some participants asked for clarification of some questions. On occasion, I asked some of the participants for additional detail or further clarification of their responses.

#### **Data Analysis**

Data analysis is an essential aspect of the research process that requires diligence and clarity in reporting. The purpose of the study influenced the data analysis technique I adopted. Because my study was a qualitative study using an interview protocol, I asked participants to confirm their understanding of the interview questions. I conducted a qualitative study to explore how knowledge retention strategies contribute to the prevention of knowledge loss in PBOs. In this qualitative study, eight participants (managers and non-managers) answered 12 semi-structured interview questions (see Appendix B).

I gathered data using semi-structured open-ended questions during interviews via Zoom. Once the audio-recorded interviews were complete, I transcribed them using a web-based audio-to-text transcription software. I reviewed the transcription several times to confirm that they were accurate. I used NVivo software to analyze the data. The data collected from the interviews contained the in-depth experiences of the study's eight

participants. The data I collected from the interview questions provided detailed information for an in-depth contextual understanding of managers and consultants' experiences in PBOs as they relate to knowledge loss and strategies to retain knowledge.

In qualitative studies, coding drives the process of the data collection, causing the researcher to restructure the instruments and the perspectives for continuing studies. Reshaping of the analysis took place during this case study, leading to themes that emerged from interviewing managers and consultants who work in PBOs. I employed two cycles of coding in the data analysis. I used descriptive coding to describe the participants' responses and pattern and focused coding to gather the codes into categories and distill them into the most meaningful themes. I repeated the coding cycles twice to ensure that I did not miss any vital themes and no new themes emerged in the data analysis process (Yin, 2014).

I started interpreting the meaning of the data based on the emergent themes and patterns (Yin, 2014). The research question was used for better understanding of the themes and patterns supported by the research findings. This data analysis approach was appropriate for qualitative research studies (Lyons & Coyle, 2016). The data analysis process was useful in uncovering the themes that answered the semi-structured research questions. Additionally, the data analysis process provided a framework to understand how knowledge retention strategies help prevent knowledge loss in PBOs. In analyzing the data, three themes and 19 codes emerged; they are outlined in Table 2.

**Table 2** *Themes and Codes* 

Themes	Codes	
Knowledge retention benefits	Increased organizational knowledge	
	Mitigate past mistakes through lessons	
	learned	
	Close operational gaps	
	Increased organizational learning	
	Operational efficiency	
Knowledge retention challenges	Lack of confidence from customers	
	Knowledge gaps	
	Knowledge hoarding	
	Organizational immaturity	
	Organizational misalignment	
	Poor leadership and management	
	Poor employee morale	
	Poor organizational culture	
Knowledge retention strategies	Communities of practice	
	Documentation	
	Cross-training	
	Job shadowing	
	Succession planning	
	Leveraging older workers	

The concepts of the KBV of the firm and IC provided a foundation for the study design to explore knowledge retention strategies that help in the prevention of knowledge loss in PBOs. Each of the concepts stresses the importance of knowledge as a strategic resource that organizations use to create value and establish a competitive advantage (Allameh, 2018; Jayasingam et al., 2016). The participants agreed that knowledge

retention is important to their respective teams and organizations, and a lack of a knowledge retention strategy influences their day-to-day business processes and projects.

### **Evidence of Trustworthiness**

## Credibility

Lemon and Hayes (2020) argued that credibility replaces internal validity. Credibility is established when the researcher has developed and articulated a certain level of confidence in the findings based on the phenomenon under investigation. To achieve credibility, I ensured adherence to the qualitative methods and instruments for this research study. Combining the proper tools (the interview protocol and member checking) and approaches utilized in case study enhanced the trustworthiness of my research (Frels & Onwuegbuzie, 2013).

Researchers use member checking to ensure that their interpretation of the event is consistent with the participants' interpretation and lived experiences (Langtree et al., 2019). I conducted member checking by giving the participants the opportunity to review their transcribed interviews. They had an opportunity to send me a response if they had any difference of opinion about what their interview said from the transcription they received. None of the participants sent back corrections or updates to their transcribed interviews. Sending transcripts to those who participated in the research study for review provided the participants an opportunity to share any concerns about my interpretations of what they shared during the interview. Agreement with the interview transcript

through member checking and a confirmation of the absence of bias in my reporting were sufficient to render my study credible.

## **Transferability**

Transferability suggests that the findings from one research study can be applied to other settings or groups of people. Daniel (2019) mentioned that transferability ensures that the content of the interviews, the behaviors, and the observed events are typical of the participants' lives. It also requires a detailed description of the context of the study, sample characteristics, participants' demographic information, participants' experiences, and the methods used to recruit the sample. The research study used a total of 8 research participants as opposed to the 24 previously intended. The 8 participants made a diverse group. Table 1 outlines the demographics of the participants. In addition to the variation in participants, detailed observational notes served as a vehicle to gather human behavioral responses to the research questions. Despite holding interviews via Zoom, observational notes collected behavioral responses that audio recordings did not easily capture. The method for ensuring transferability, called the thick description, is a valid method to enhance the probability of transferability (Serra, 2016).

## **Dependability**

Morse (2015) explained that dependability is the ability to obtain the same results if the study were to be repeated. Dependability is evaluated by considering the decisions made and steps taken during the research process (Langtree et al., 2019). To establish dependability for this research study, I developed an audit trail on the data for this

research, including participant communication, how the collection activity occurred, when the participants provided the data, and the tools I used to clean the data and then analyze the data. Audit trails provide a method for increased dependability and trustworthiness of qualitative data findings (Sinkovics & Alfoldi, 2012).

In addition to the audit trail for data collection, the data analysis steps were clearly outlined with the first pass of the data analysis to find descriptive coding nodes. There was a second coding cycle to identify additional themes that emerged after reviewing the descriptive coding nodes. In a few cases, I did an additional coding cycle to clarify where information related to the second cycle's themes might emerge.

## Confirmability

Confirmability is equated with the objectivity of the phenomenon under investigation and addresses whether the interpretations and findings are from the participants' lived experiences and do not include the researcher's biases (Lemon & Hayes, 2020). The primary element of confirmability I used for this study was the concept of reflexivity in qualitative research. All of the interviews were transcribed and sent to the participants for review and validation. None of the participants had any concerns or issues with their respective transcripts and validated their responses to the interview questions.

My experience as a full-time employee who worked in a PBO provided me with a significant component of reflexivity. This component included the researcher-researched relationship and the researcher's worldview (Berger, 2015). Having worked on a project

within a PBO, I had professional experience with events that occurred when management did not have a knowledge retention strategy to implement on the project, and project knowledge was lost.

## **Study Results**

The results of this research study could help clarify understanding of how knowledge retention strategies help to prevent knowledge loss in PBOs. My study addressed the research question and the various knowledge retention strategies used by managers in PBOs to prevent knowledge loss. After reviewing the data collected from interviews, observation notes, and member-checked forms, three themes emerged with respect to the research question (Table 2). The research question is as follows: What knowledge retention strategies do managers use to prevent knowledge loss? The participants and supporting documents provided rich data on the knowledge retention strategies used to prevent knowledge loss. Identifying the perception and benefits of knowledge retention and the strategies used by managers formed the major aspect of the study. The following subsections describe the participant's sample, the five themes that emerged, and how the data collected supported the central research question, aligned with existing research, and validated the choices of the two concepts that formed the conceptual framework for this study.

### **Theme 1: Knowledge Retention Benefits**

The first theme makes clear the participants' perceptions of knowledge retention.

Research findings suggest knowledge retention is beneficial to managers and their teams.

Most of the research participants affirmed that knowledge retention is important to maintain business and operational efficiency, improve process delivery, and close operational gaps. Participant M1 mentioned that knowledge retention is significant because it helps him make sure his team members can utilize all of the knowledge available to complete their tasks quicker. Participant M3 stated that knowledge retention is critical as it relates to performing operations efficiently and effectively. Participant C1 commented, "Knowledge retention helps people get up to speed quicker." This theme that emerged from the data helped form one of the significant aspects of the study. The research data indicates that knowledge retention is critical for managers to keep continuity. Additional supporting statements from the participants are shown in Table 3.

**Table 3**Theme 1: Knowledge Retention Benefits

Participant	Participant comments
C1	It would instantly allow people to get up to speed quicker. Then potentially falling through may be old manuals because nobody puts anything in manuals anymore. It would let people get up to speed quickly and hit the ground running after somebody leaves, be able to transfer that knowledge to new people who may have never come in contact with a person, and it just gives, I would imagine, that it would give the organization a sense of security they will not lose, you know, potentially the type of knowledge that isn't, you know standard, you know, because there's always standards of doing everything, but then ways things are done just by doing them every day different ways that people come up to, I don't want to say cut corners which is a different methodology outside of you know,

the standard textbook way of doing things, you know, that people, you know gained through doing the actual work, you know.

That's very highly important to me as a manager because I need to know that my team members that I have can consistently follow whatever the knowledge they have, whether it's from the job or from something else, that it can flow through the procedures that they have and they can utilize that knowledge to, you know, complete their tasks easier.

I think the biggest benefit is that it reduces dependency on a single person. The—which tends not to be the healthiest team atmosphere to have more ideas and more knowledge, but also if something happens to that person or they leave, it's bad for the project, increases overall efficiency and effectiveness and to it even when it's not related to people leaving the firm or something like that, knowledge transfer helps promote healthy growth, and transfer of responsibilities are separate areas within the firms of people able to move up and take on more responsibility in a cyclical fashion.

I feel like knowledge retention is very important, especially as it relates to a specific topic or operational function that a specific team performs.

The benefit of knowledge retention would be to prevent making the consistent missteps and, and overall mismanagement of funds. Many times individuals are learning critical lessons that can improve process delivery. As you're thinking about not just process, but product delivery as well, and as you're thinking about reproduction of effort, it costs the company more and more money just to correct past issues as opposed to being able to just mitigate it in the future. You know, just taking the information that you've learned, understanding what your mistakes have been and don't repeat them in the future.

M1

M2

M3

M4

## **Theme 2: Knowledge Retention Challenges**

The second theme makes clear the challenges participants have with knowledge retention in their organization. Most of the research participants spoke freely about the challenges their organizations have when knowledge retention is not a priority. Some of the challenges the participants discussed were team members hoarding knowledge and organizational leadership not encouraging knowledge retention. Participant C1 talked about experiences meeting older workers who did not want to share their knowledge with younger workers. Participant M2 explained that a lack of a knowledge retention strategy could cause inefficiency. Participant M3 shared an experience of managers being "territorial" with their knowledge. The research data indicates that when there are challenges to retain knowledge, loss of knowledge and other factors that impact the competitive advantage of an organization will occur. Additional supporting statements from the participants are listed in Table 4.

**Table 4**Theme 2: Knowledge Retention Challenges

Participant	Participant comments
C1	I will not say many, but yes, I have come across people, and it's terrible to say it's usually older people tend to want to hold on to it. I don't know if they feel as the workforce gets younger that the younger crowd is coming in to take their place at least before they retire, and they don't want

their job, their livelihood put at risk, so they hold on to that knowledge just a little bit tighter because it secures their position within the company or the agency.

In my current organization there is definitely a lack of knowledge retention. I think the main driver for businesses for the last 10 years has been financial cuts, reducing expenses.

If you don't have knowledge retention, they (customers, clients) will get annoyed. They may lack the confidence in you as a consultant and it will lead to inefficiency too, in addition to just their perception about it, it will result in less positive experiences with you.

I think knowledge retention is very important. I think that organizations, especially immature organizations or people that are very matrix and they make decisions from high levels, they tend not to fully investigate how the implications of making decisions like reorgs, right. Perhaps they don't fully understand the implications of letting John go, who is three levels down when Johnny is the only person that's really been performing a certain function for the last five years and nobody else knows how to do it. He hasn't documented that anywhere. So, I think overall high level, the important thing to do, and I realize, I mean, I think one of the resistance factors to this is people are afraid to share information because they're insecure.

So, I can tell you that I learned from a hard lesson learned. I was, as I first joined my current organization, there was an employee that had been there for 27 years, and so just imagine the knowledge loss when an individual walked out the door abruptly, no two weeks' notice, no nothing. So, when I heard that he was leaving, I kept bringing it to the attention of upper management, asking that they you know, would work with us to try to get that knowledge transfer documented and some type of evidence of proof, right? Like we needed to get from him and it was shrugged off. Like it wasn't a big deal. We're better off without him. You know, we don't really need them, you know, and I was actually taken aback from that.

C2

M2

M3

M4

## **Theme 3: Knowledge Retention Strategies**

The third theme makes clear the participants' understanding of different knowledge retention strategies they use. The participants described the importance of using documentation, developing a community of practice, job shadowing, and cross training as strategies to retain knowledge on their teams and organizations to ensure continuity and avoid knowledge gaps. Participant M1 emphasized how documentation removes the many challenges of knowledge sharing within an organization. Participant C1 described a situation where she documented her existing knowledge to help a new team member get up to speed on the current business and operational processes. The research data indicates that different strategies are implemented by managers to retain knowledge on projects, teams, and organizations. Additional supporting statements from the participants are shown in Table 5.

**Table 5**Theme 3: Knowledge Retention Strategies

Participant	Participant comments
C1	You know that the last person that joined our team was probably maybe two years ago, and some of the things that I did beforehand was put together an email that pretty much laid out everything that he would need access to, you know, to start working immediately, which encompasses I do anyone cured earnest and

getting people's account setup and things like that. So, he had an email that outlined that for him.

M1

So, a couple of things right number one is documentation. Everybody needs to document. Not only what they do on a monthly basis, but they need to be able to train another team member to be able to do that so that way that continuity stays there. So that's super important to me.

I've always taken pride in having cross-training and that's one thing that we never had. You know where I'm at right now is that we always had to you know, everybody was siloed. I should say everybody was doing one thing, and the other hand didn't know what that person was doing. So it was important to have cross-training so that way should somebody leave or not show up. They get sick whatever it is and can't come back at least. Somebody knows something about the process.

So for the more explicit knowledge formally documenting roles and responsibilities and providing historical project documentation, for context, kind of two different outputs there, and then to help gain that more tacit knowledge I think informal Q and A's between ideally the person transitioning out and transitioning in where they can kind of debrief and say, what are we doing and why, getting feedback and the most important thing.

So, one I've used in the past that was actually pretty nifty, we had created a webinar for a project for the client to share with their employees. I was able to share the webinar with the person coming on. It explained the entire project. It's not something we would normally create, but we do have Zoom recording capability. I think we could. So, there was that and then documentation review, of course I've mentioned in addition to the informal Q & A, I think just the act of observing someone else, attending a meeting where my only job is if I'm the person coming on, it's just to pay

M2

attention and see what questions come up in that interactive, actual environment compared to just reading documents or hearing about it.

M3

So, the strategy I use is to document knowledge. So, when we have something that somebody knows and that somebody else may not know, I always say, make a SharePoint, do a PowerPoint. Let's save it to a collective area where we can all access that information. Right, it's very important in, in my field to preserve that information, because if you don't have it, you're not going to operationalize correctly.

M4

And then you have those not, not so formal conversations where people just say, well, you know, this is what happened before. So, a lot of times a lot of conversations start, well, historically this, you know, we've seen this with this case in this customer. So, it's a lot of conversation, but it may not always be formally documented, but I have tried to start to impose that formal documentation of those things so that we can make sure that we're capturing lessons learned, which is something our organization moves way too fast to even go back and do like, or post-mortem 30-, 60-, 90-day lessons learned.

M5

The main thing that I've tried to do is when I have someone that's doing a particular role that I try and make sure that they document what it is that they do. And so, by them documenting it because they're actually doing the job. And then we go through that process of understanding exactly what that job entails, whether it looks the same as I intended it, or it's evolved over time that I try and make sure that I look at that as something that I can have in bed, if they were to leave that they can lead that information with us.

# **Summary**

The research question I answered in the presentation of findings for this research study was, What knowledge retention strategies do managers use to prevent knowledge loss? The study's findings revealed that there are different knowledge retention strategies that managers use in day-to-day operations. Additionally, the study's findings revealed the benefits and challenges of knowledge retention to managers and organizations. Chapter 5 includes the interpretation of the findings, limitations of the qualitative research study, recommendations, implications for positive social change, and conclusion.

## Chapter 5: Discussion, Conclusions, and Recommendations

The purpose of this case study was to explore knowledge retention strategies to prevent knowledge loss in PBOs. This study included semi-structured interviews of eight experienced managers and consultants. This chapter presents a detailed explanation of the findings and limitations of the study. Additionally, this chapter addresses recommendations for professional practice as well as future studies, implications for social change, and the conclusion of the study.

## **Interpretation of Findings**

Knowledge retention has remained an essential strategy in preventing knowledge loss and ensuring project and business continuity. The interview responses indicated that knowledge retention is an excellent way of ensuring that organizations have continuity in key team roles without knowledge loss. The participants agreed with Ensslin et al.'s (2020) assertion that knowledge retention helps formalize an organization's commitment to its long-term strategic objectives. Makhubela and Ngoepe (2018) similarly stated that knowledge retention is a solution to combating knowledge loss by accumulating, maintaining, and identifying knowledge lost by the organization. Retaining that knowledge can help the organization maintain a competitive advantage.

All participants in this research study suggested that a knowledge retention strategy helps managers maintain business continuity and efficiency. Chigada and Ngulube (2016) stressed that managers face a challenging task to preserve organizational knowledge, which entails designing and implementing a knowledge retention strategy.

Managers should implement a knowledge retention strategy proactively in anticipation of when an employee will leave unexpectedly or take early retirement. Makhubela and Ngoepe (2018) argued that a knowledge retention strategy helps improve innovation, organizational growth, employee development, and business efficiency. Levy (2011) explained that a knowledge retention strategy helps control business losses and avoid starting over within organizations.

Acharya and Mishra (2017) viewed a knowledge retention strategy as a solution to help managers reuse or reapply essential knowledge in the future to propose innovative solutions to newer problems and even train other employees on project engagements with important clients. Motshegwa (2017) added that implementing a knowledge retention strategy requires that organizations be aware of factors that can enhance or impede knowledge retention. The study's findings indicated that managers see benefits from knowledge retention and the importance of a knowledge retention strategy to maintain business continuity and efficiency.

The findings also indicated challenges with knowledge retention. First, some managers and organizations did not emphasize knowledge retention. Although there is an emphasis on knowledge retention, managers tend to have little interest in this issue, and few organizations have a formal, well-functioning knowledge retention strategy in place (Wikström et al., 2018). Second, some of the older employees tended to hold on to their knowledge and not share with team members. Some of the participants commented on experiences with team members who declined to share their knowledge, such as

Participant M3, who described the behavior of some employees who would not share knowledge as territorial. Anand et al. (2020) added that while organizations put a lot of effort into cultivating knowledge-sharing activities among employees, the bottom line is that the success of these efforts relies on the employees' willingness to share their knowledge. Some participants reassured employees who were hoarding knowledge that they were not a threat and convinced them to share their knowledge with the team.

Another challenge to knowledge retention is employee turnover. All participants in this research study confirmed that employee turnover, whether voluntary or involuntary, does have an effect on knowledge retention. Kanade et al. (2015) asserted that employee turnover costs organizations thousands of dollars annually. Martins and Meyer (2012) commented that employee turnover leaves huge gaps in valuable knowledge that are difficult to identify until unexpected quality problems, mistakes, costly disruptions in performance or operations, loss of competitive advantage, and even tragic accidents occur. Johansen (2013) commented that employee turnover requires the organization to invest significant resources to recruit, interview, train, and socialize new workers.

The findings from this study affirmed that there are different knowledge retention strategies managers and employees use to prevent knowledge loss. The participants discussed using different strategies, such as a) communities of practice, b) documentation, c) cross-training, d) job shadowing, and e) succession planning. Patriotta et al. (2013) argued that the transfer and sharing of knowledge and best practices with the

rest of the organization increases the organization's ability to exploit its knowledge base and improve performance.

Only one of the eight participants talked about developing a community of practice to share and retain knowledge. Communities of practice are beneficial to driving strategy in organizations (Manuti et al., 2017). Bratianu (2019) stated that communities of practice stimulate knowledge sharing. A key advantage for communities of practice is the environment of trust created that reduces the effects of knowledge hoarding.

Documentation was a knowledge retention strategy affirmed by most of the research participants. Levallet and Chan (2019) asserted that knowledge retention increases when documented knowledge is available for other employees to use. New team members use documented tacit knowledge to understand existing processes and reduce the risk of knowledge loss.

The findings from this study confirmed the use of succession planning as a knowledge retention strategy. Participant M2 discussed using a succession plan when she transferred to another project. Chigada and Ngulube (2016) asserted that succession plans are an orderly knowledge retention strategy for business and project survival. Sabir and Kalyar (2013) added that taking policy initiatives that would promote organizational learning through enhanced business practices such as succession planning could result in bridging the knowledge gap and increasing competitive advantage, employee retention, and job satisfaction.

## **Limitations of the Study**

There were a couple of limitations presented in this study. The first limitation for this study was the potential for researcher bias arising from my personal views on knowledge loss and knowledge retention and its impact on PBOs. My knowledge of the phenomenon and the social setting presented a potential bias. To prevent this bias, I avoided asking leading questions or preempting the participants' answers. Additionally, I utilized bracketing, as recommended by Onwuegbuzie and Byers (2014), to reduce personal bias.

Another limitation of this study was the actual sample size. The intended sample size was 24 participants. I initially used purposeful sampling and social media posts directed at professional organizations to recruit participants. However, I was only able to receive responses from eight participants. The number of responses can be viewed as low and potentially limited the study from having more diverse viewpoints.

#### Recommendations

The first recommendation for future research is to extend the study to include project executives and directors who work in PBOs. The participants in this study were a combination of managers, full-time employees, and consultants. Ensslin et al. (2020) stated that executives have a hard time understanding how well their processes and procedures promote knowledge retention. Extending the study to include project executives might provide an additional perspective on knowledge retention and strategies to mitigate knowledge loss.

The second recommendation for future research is to examine more in-depth other knowledge retention strategies such as mentoring. Chigada and Ngulube (2016) identified mentoring as a knowledge retention strategy where senior managers transfer their knowledge to less experienced colleagues in a short period. Wheeler and Cooper (2016) viewed mentoring as a strategy to support knowledge retention, succession planning, and job satisfaction. Ross (2013) stated that an excellent mentoring program focuses on organizational purpose and is streamlined to meet the organizational goals. More research into mentoring and other knowledge retention strategies not identified in this research study is needed.

The third recommendation for future research is to study how an organization's culture affects knowledge retention. Organizational culture is essential to knowledge retention (Jayawickrama et al., 2019). Sitlington and Marshall (2011) asserted that organizations should consider their culture and climate regarding the mechanisms for knowledge retention. Some of the research participants noted that knowledge retention was not a focal point within their organization. More research is needed to understand how an organization's culture influences knowledge retention. Ma et al. (2014) concluded that a trusting culture improves knowledge retention, and its leaders and managers should try to create such a culture.

The fourth recommendation for future research is to study the relationship between leadership and knowledge retention. The role of leadership is to promote a positive cultural direction towards knowledge retention (Yang et al., 2014). Previous

research studies looked at the relationship between transformational leadership and knowledge sharing (Anselmann & Mulder, 2020; Pellegrini et al., 2020). Further research into the relationship of other leadership types (transactional, servant, charismatic, and coaching) and knowledge retention is warranted.

The fifth recommendation for future research is to study the effect of knowledge hoarding and its impact on knowledge retention strategy. Anand et al. (2020) stated that, despite investing in knowledge sharing facilitation, knowledge hoarding remains prevalent in organizations. When employees withhold information from each other, it has a long-term negative impact on trusting each other (Holten et al., 2016). Several of the research participants shared examples where they encountered another team member who hoarded their knowledge and how they gained the team member's trust in order for them to share their knowledge. De Geofroy and Evans (2017) agreed that increased trust among employees is important to reduce knowledge hoarding. Organizations can help to limit knowledge hoarding by encouraging teamwork and collaboration.

#### **Implications**

# **Contribution to Individuals**

The significance to the employees of PBOs is that establishing a knowledge retention strategy has the distinct advantage of improving efficiency and organizational growth, enhancing employee development and competitive advantage. The participants revealed that knowledge retention closes operational gaps, keeps continuity, and is critical to an organization's success. Gaghman (2019) noted that employees are more

willing to contribute to knowledge sharing in a positive workplace environment. An employee's positive attitude can highly influence their contribution to knowledge transfer, sharing, and retaining practices.

#### **Contribution to Business Practice**

The turnover rate of skilled workers has increased throughout history. From the 1960s to the 2000s, the turnover rate ranged from 15% to 28% (Lo, 2015). Lee et al. (2018) argued that turnover is detrimental to organizations in terms of higher costs for recruitment and training, loss of organizational memory, and decreasing productivity. Kim et al. (2013) stated that the shortage of skilled workers would increase as the global workforce gets older and fewer younger workers are available to meet the labor demand. The trademark of a successful organization depends on the degree to which it generates, maintains, and protects knowledge.

Sumbal et al. (2018) noted that when there is loss of critical skilled workers there is a risk of knowledge loss, which could be the foundation for the competitive advantage of an organization. Therefore, it is critical to identify the important knowledge possessed by departing employees and reacting applicably to retain that knowledge. The findings from this study on exploring knowledge retention strategies to prevent knowledge loss showed that formal knowledge retention strategies need to be developed and supported by managers throughout the organization. Project-based organizations should have a knowledge retention strategy to retain organizational knowledge and mitigate the risk of knowledge loss. Ramona and Alexandra (2019) affirmed that even though the process of

retaining knowledge in organizations is considered crucial for long-term business success and maintaining a competitive edge, only a few organizations have clear and formal knowledge retention strategies.

The implication for positive social change includes the potential to impact project teams across different industries by contributing to the enhancement of knowledge retention strategies. The continuing sharing of knowledge among teammates would add to the organization's knowledge repository and lessen knowledge gaps while ensuring business continuity. The findings from the qualitative study participant interviews revealed that knowledge sharing and knowledge transfer practices are crucial to employee development and job performance. Ahmad and Karim (2019) suggested that knowledge retention has a positive effect on employee performance, increased innovation in the work place, and an improved team climate. Organizations, including PBOs, could use the qualitative study results to develop and formalize existing knowledge retention strategies and increase their adequacy.

#### **Contributions to Society**

An organization's formal knowledge retention strategy is important for long-term success. The overall strategy needs to have knowledge sharing and knowledge transfer strategies that can address any knowledge gaps that may occur. Managers need to create an environment where knowledge sharing is encouraged. The qualitative study participants confirmed that knowledge retention strategies are critical to training and continuing employee development within the team and organization. These strategies

should strive to develop individuals with the necessary technical and business knowledge for when employee turnover happens. Moreover, retaining critical knowledge also prevents organizations from starting from scratch every time a project gets underway. In some cases, these projects create products that benefit society.

# **Contribution to Theory**

The study findings revealed that there are not enough formal knowledge retention strategies in PBOs to prevent knowledge loss. The qualitative approach adopted for this research study provided a solid basis for initial data collection that would inform further research on the relationship between knowledge retention and knowledge loss. According to Mayan (2016), qualitative researchers work inductively from individual cases and a preexisting framework or a particular theory.

The findings from the qualitative study filled a gap in the literature by providing knowledge about knowledge retention strategies to prevent knowledge loss in PBOs.

Additionally, the qualitative study findings provided insights regarding how documentation, collaboration, role shadowing, and cross training are some of the knowledge retention strategies that are used by managers to help mitigate knowledge loss.

I recommend developing a knowledge retention strategy that includes the following steps: (1) identification of critical knowledge, (2) assessing the risk of losing the critical knowledge identified, and (3) actions to preserve the critical knowledge. I also recommend managers work with their team to develop knowledge sharing practices such

as cross-training and informal training sessions. These practices will help to transform individual knowledge into organizational knowledge. Additionally, I recommend providing incentives for staff who actively participate in sharing their knowledge with the team. Employees should feel the value of sharing their ideas and knowledge with their teammates and understand that their input is important to the overall performance of the team.

#### **Conclusions**

The emergent themes reveal how knowledge retention strategies supplemented by knowledge sharing and knowledge transfer practices in PBOs might contribute to preventing knowledge loss. Feedback from the research participants provided valued information on how knowledge retention strategies can help to prevent knowledge loss within their organization. It was noted by the participants that knowledge retention prevents knowledge loss by implementing knowledge sharing and knowledge transfer practices to reduce employee turnover.

Organizations face increasing worldwide competition due to globalization. This reality makes them aware of the need to see the knowledge and expertise of their employees as a critical asset (Ramona & Alexandra, 2019). Implementing effective knowledge retention strategies would have a lasting effect on the employees, business success, and culture and benefit both the organization and society. Managers have an important role in supporting the implementation of knowledge retention strategies.

Daghfous et al. (2013) noted that managers should act as role models for employees and

be proactive in their support. Similarly, Liu et al. (2020) asserted that managers should maintain high spirits and a joyful atmosphere in the workplace to motivate people to share knowledge.

The results contribute to the existing body of knowledge regarding how organizations use knowledge retention strategies for business continuity. An organization's failure to incorporate knowledge sharing practices such as cross-training and documentation could result in knowledge loss and create knowledge gaps. Effective knowledge retention strategies may mitigate knowledge loss and increase decision-making by ensuring that knowledge is readily available to employees. By highlighting the importance of knowledge retention, this research study raised the awareness of the importance of employees and the value their knowledge brings to their respective organizations.

#### References

- Acharya, A., & Mishra, B. (2017). Exploring the relationship between organizational structure and knowledge retention: A study of the Indian infrastructure consulting sector. *Journal of Knowledge Management*, 21(4), 961–985.

  https://doi.org/10.1108/JKM-11-2016-0506
- Adams, E. (2010, July). The joys and challenges of semi-structured interviewing.

  \*Community Practitioner, 83(7), 18–21.
- Addis, M. (2016). Tacit and explicit knowledge in construction management.

  \*Construction Management and Economics, 34(7-8), 439–445.

  https://doi.org/10.1080/01446193.2016.1180416
- Agostini, L., & Nosella, A. (2017). A dual knowledge perspective on the determinants of SME patenting: Results of an empirical investigation. *Management Decision*, 55(6), 1226–1247. https://doi.org/10.1108/MD-04-2016-0215
- Ahmad, F., & Karim, M. (2019). Impacts of knowledge sharing: A review and directions for future research. *Journal of Workplace Learning*, *31*(3), 207–230. https://doi.org/10.1108/JWL-07-2018-0096
- Ajmal, M., Helo, P., & Kekale, T. (2010). Critical factors for knowledge management in project business. *Journal of Knowledge Management*, *14*(1), 156–168. http://dx.doi.org/10.1108/13673271011015633
- Ajmal, M., & Koskinen, K. (2008). Knowledge transfer in project-based organizations:

  An organizational culture perspective. *Project Management Journal*, *39*(1), 7–15.

## http://dx.doi.org/10.1002/pmj.20031

- Akanbi, P. A. (2016). Exploring the link between intellectual capital and perceived organizational performance. *International Journal of Information, Business and Management*, 8(2), 35–45.
- Akhavan, P., Zahedi, M. R., & Hosein, S. H. (2014). A conceptual framework to address barriers to knowledge management in project-based organizations. *Education*, *Business and Society: Contemporary Middle Eastern Issues*, 7(2/3), 98–119. https://doi.org/10.1108/EBS-10-2013-0040
- Alavi, M., & Leidner, D. E. (2001, March). Knowledge management and knowledge management systems: Conceptual foundations and research issues. *MIS Quarterly*, 25(1), 107–136.
- Ali, I., Musawir, U., & Ali, M. (2018). Impact of knowledge sharing and absorptive capacity on project performance: The moderating role of social processes. *Journal of Knowledge Management*, 22(2), 453–477. <a href="https://doi.org/10.1108/JKM-10-2016-0449">https://doi.org/10.1108/JKM-10-2016-0449</a>
- Allameh, S. M. (2018). Impact of knowledge sharing and absorptive capacity on project performance: The moderating role of social processes. *Journal of Intellectual Capital*, 19(5), 858–874. <a href="https://doi.org/10.1108/JIC-05-2017-0068">https://doi.org/10.1108/JIC-05-2017-0068</a>
- Almeida, M. V., & Soares, A. L. (2014). Knowledge sharing in project-based organizations: Overcoming the informational limbo. *International Journal of Information Management*, *34*(1), 770–779.

## https://doi.org/10.1016/j.ijinfomgt.2014.07.003

- Almudallal, A. W., Muktar, S. N., & Bakri, N. (2016). Knowledge management in the Palestinian higher education: A research agenda. *International Review of Management and Marketing*, 6(S4), 91–100.
- Amayah, A. T. (2013). Determinants of knowledge sharing in a public sector organization. *Journal of Knowledge Management*, *17*(3), 454–471. https://doi.org/10.1108/JKM-11-2012-0369
- Anand, A., Centobelli, P., & Cerchione, R. (2020). Why should I share knowledge with others? A review-based framework on events leading to knowledge hiding.

  \*\*Journal of Organizational Change Management, 33(2), 379–399.\*\*

  https://doi.org/10.1108/JOCM-06-2019-0174
- Anderson, C. (2010). Presenting and evaluating qualitative research. *American Journal of Pharmaceutical Education*, 74(8), 1–7.
- Anselmann, V., & Mulder, R. H. (2020). Transformational leadership, knowledge sharing and reflection, and work teams' performance: A structural equation modelling analysis. *Journal of Nursing Management*, 28(7), 1627–1634.

  <a href="https://doi.org/10.1111/jonm.13118">https://doi.org/10.1111/jonm.13118</a>
- Argote, L. (2013). Organizational Learning: Creating, retaining and transferring knowledge (2nd ed.). Springer Science+Business Media.
- Argyris, C., & Schon, D. (1978). Organizational learning: Creating, retaining and transferring knowledge. Addison-Wesley.

- Aribi, A., & Dupouet, O. (2016). Absorptive capacity: A non-linear process. *Knowledge Management Research & Practice*, 14(1), 15–26.

  <a href="https://doi.org/10.1057/kmrp.2014.17">https://doi.org/10.1057/kmrp.2014.17</a>
- Arora, R. (2016, Winter). Strategic role of HR in efficient and effective management of the human capital pool. *Management Review: An International Journal*, 11(2), 23–42.
- Arora, R. G. (2015, April-June). Comparative study of factors affecting talent retention and strategic remedies. *Anvesha*, 8(2), 10–20.
- Arsenijevic, O., Trivan, D., Podbregar, I., & Sprajc, P. (2017, May). Strategic aspect of knowledge management. *Organizacija*, 50(2), 163–177.

  <a href="https://doi.org/10.1515/orga-2017-0011">https://doi.org/10.1515/orga-2017-0011</a>
- Artto, K., Valtakoski, A., & Kärki, H. (2015). Organizing for solutions: How project-based firms integrate project and service businesses. *Industrial Marketing Management*, 45(2), 70–83. <a href="https://doi.org/10.1016/j.indmarman.2015.02.021">https://doi.org/10.1016/j.indmarman.2015.02.021</a>
- Asiaei, K., & Jusoh, R. (2015). A multidimensional view of intellectual capital: The impact on organizational performance. *Management Decision*, *53*(3), 668–697. <a href="https://doi.org/10.1108/MD-05-2014-0300">https://doi.org/10.1108/MD-05-2014-0300</a>
- Asrar-ul-Haq, M., & Anwar, S. (2016). A systematic review of knowledge management and knowledge sharing: Trends, issues, and challenges. *Cogent Business & Management*, *3*(1), 1–17. https://doi.org/10.1080/23311975.2015.1127744
- Aydin, E., & Gormus, A. S. (2015). Does organizational forgetting matter?

- Organizational survival for life coaching companies. *The Learning Organization*, 22(3), 150–162. http://dx.doi.org/10.1108/TLO-12-2014-0068
- Ayub, Y. I., Kogeda, O. P., & Lall, M. (2018). Capturing tacit knowledge: A case of traditional doctors in Mozambique. South African Journal of Information
  Management, 20(1), 1–8. <a href="https://doi.org/10.4102/sajim.v20i1.880">https://doi.org/10.4102/sajim.v20i1.880</a>
- Baillie, L. (2015). Promoting and evaluating scientific rigour in qualitative research.

  Nursing Standard, 29(46), 36–42. https://doi.org/10.7748/ns.29.46.36.e8830
- Bairi, J., Manohar, B. M., & Kundu, G. K. (2011). Knowledge retention in the IT service industry. *Journal of Systems and Information Technology*, *13*(1), 43–65. https://doi.org/10.1108/13287261111118340
- Banerjee, D., & Ray, A. (2016). Knowledge management as predictor of transformational leadership: An empirical observation on middle managers of IT companies of Kolkata. *International Journal of Education & Management Studies*, 6(2), 144–148. www.iahrw.com/index.php/home/journal\_detail/21#list
- Barkat, W., & Beh, L. (2018). Impact of intellectual capital on organizational performance: Evidence from a developing country. *Academy of Strategic Management Journal*, 17(2), 1–8. <a href="https://www.semanticscholar.org/paper/Impact-of-Intellectual-Capital-on-Organizational-a-Barkat-Beh/ea3814e5da909b6b8c320ad4216b48a2d339e339">https://www.semanticscholar.org/paper/Impact-of-Intellectual-Capital-on-Organizational-a-Barkat-Beh/ea3814e5da909b6b8c320ad4216b48a2d339e339</a>
- Barnham, C. (2015). Quantitative and qualitative research: Perceptual foundations. *International Journal of Market Research*, 57(6), 837–854.

## https://doi.org/10.2501/IJMR-2015-070

- Baron, A. (2011). Measuring human capital. *Strategic HR Review*, *10*(2), 30–35. http://dx.doi.org/10.1108/14754391111108338
- Bartsch, V., Ebers, M., & Maurer, I. (2013). Learning in project-based organizations: The role of project teams' social capital for overcoming barriers to learning.

  \*International Journal of Project Management, 31(1), 239–251.\*

  https://doi.org/10.1016/j.ijproman.2012.06.009
- Bashouri, J., & Duncan, G. W. (2014). A model for sharing knowledge in architectural firms. *Construction Innovation*, *14*(2), 168–185. <a href="https://doi.org/10.1108/CI-10-2012-0057">https://doi.org/10.1108/CI-10-2012-0057</a>
- Baškarada, S. (2014). Qualitative case study guidelines. *The Qualitative Report*, 19(40), 1–18. http://nsuworks.nova.edu/tqr/vol19/iss40/3
- Berger, R. (2015). Now I see it, now I don't: Researcher's position and reflexivity in qualitative research. *Qualitative Research*, *15*(2), 219–234. https://doi.org/10.1177/1468794112468475
- Berry, L. E. (2016). The research relationship in narrative enquiry. *Nurse Researcher*, 24(1), 10–14. https://doi.org/10.7748/nr.2016.e1430
- Bessick, J., & Naicker, V. (2013). Barriers to tacit knowledge retention: An understanding of the perceptions of the knowledge management of people inside and outside the organization. *South African Journal of Information Management*, 15(2), 1–8. <a href="http://dx.doi.org/10.4102/sajim.v15i2.556">http://dx.doi.org/10.4102/sajim.v15i2.556</a>

- Bharati, P., Zhang, W., & Chaudhury, A. (2015). Better knowledge with social media?

  Exploring the roles of social capital and organizational knowledge management.

  Journal of Knowledge Management, 19(3), 456–475.

  https://doi.org/10.1108/JKM-11-2014-0467
- Birt, L., Scott, S., Cavers, D., Campbell, C., & Walter, F. (2016). Member checking: A tool to enhance trustworthiness or merely a nod to validation? *Qualitative Health Research*, 26(13), 1802–1811. <a href="https://doi.org/10.1177/1049732316654870">https://doi.org/10.1177/1049732316654870</a>
- Bloomberg, L. D., & Volpe, M. (2012). Completing your qualitative dissertation: A roadmap from beginning to end. Sage.
- Boddy, C. R. (2016). Sample size for qualitative research. *Qualitative Market Research:*An International Journal, 19(4), 426–432. <a href="https://doi.org/10.1108/QMR-06-2016-0053">https://doi.org/10.1108/QMR-06-2016-0053</a>
- Bourouni, A., Noori, S., & Jafari, M. (2014). Organizational groupings and performance in project-based organizations: An empirical investigation. *Aslib Journal of Information Management*, 66(2), 156–174. <a href="https://doi.org/10.1108/AJIM-05-2013-0049">https://doi.org/10.1108/AJIM-05-2013-0049</a>
- Bourouni, A., Noori, S., & Jafari, M. (2015). Knowledge network creation methodology selection in project-based organizations: An empirical framework. *Aslib Journal of Information Management*, 67(1), 74–93. <a href="http://dx.doi.org/10.1108/AJIM-08-2014-0106">http://dx.doi.org/10.1108/AJIM-08-2014-0106</a>
- Brajer-Marczak, R. (2016). Elements of knowledge management in the improvement of

- business processes. *Management*, 20(2), 242–260. https://doi.org/10.1515/manment-2015-0063
- Brandi, U., & Iannone, R. L. (2015). Innovative organizational learning technologies:

  Organizational learning's Rosetta Stone. *Development and Learning in*Organizations, 29(2), 3–5. <a href="https://doi.org/10.1108/DLO-07-2014-0048">https://doi.org/10.1108/DLO-07-2014-0048</a>
- Bratianu, C. (2018). A holistic approach to knowledge risk. *Management Dynamic in the Knowledge Economy*, 6(4), 593–607. https://doi.org/10.25019/MDKE/6.4.06
- Bratianu, C. (2019). Exploring knowledge entropy in organizations. *Management Dynamic in the Knowledge Economy*, 7(3), 353–366.

  <a href="https://dx.doi.org/10.25019/MDKE/7.3.05">https://dx.doi.org/10.25019/MDKE/7.3.05</a>
- Brown, R. S. (2014). Researcher cognition and the effect of the external environment on business scholars. *Academy of Marketing Studies Journal*, 18(2), 227–237.
- Bryant, A., & Charmaz, K. (2012). *The SAGE handbook of grounded theory*. SAGE Publishing.
- Bungay, V., Oliffe, J., & Atchison, C. (2016). Addressing underrepresentation in sex work research: Reflections on designing a purposeful sampling strategy.

  \*\*Qualitative Health Research\*, 26(7), 966–978.\*\*

  https://doi.org/10.1177/1049732315613042
- Butina, M. (2015, Summer). A narrative approach to qualitative inquiry. *Clinical Laboratory Science*, 28(3), 190–196.
- Cameron, R., & Molina-Azorin, J. F. (2014). The acceptance of mixed methods in

- business and management. *International Journal of Organizational Analysis*, 22(1), 14–29. https://doi.org/10.1108/IJOA-08-2010-0446
- Candela, A. G. (2019). Exploring the function of member checking. *The Qualitative Report*, 24(3), 619–628. https://nsuworks.nova.edu/tqr/vol24/iss3/14
- Cao, Y., & Xiang, Y. (2012). The impact of knowledge governance on knowledge sharing. *Management Decision*, 50(4), 591–610. http://dx.doi.org/10.1108/00251741211220147
- Castaneda, D. I., Manrique, L. F., & Cuellar, S. (2018). Is organizational learning being absorbed by knowledge management? A systematic review. *Journal of Knowledge Management*, 22(2), 299–325. <a href="https://doi.org/10.1108/JKM-01-2017-0041">https://doi.org/10.1108/JKM-01-2017-0041</a>
- Chauhan, V. S., & Patel, D. (2013, February). Employee turnover: A factorial study of IT industry. *Journal of Strategic Human Resource Management*, 2(1), 37–44.
- Chigada, J., & Ngulube, P. (2016). A comparative analysis of knowledge retention strategies at selected banks in South Africa. *Business Information Review*, *33*(4), 221–227. <a href="https://doi.org/10.1177/0266382116683892">https://doi.org/10.1177/0266382116683892</a>
- Chitsazan, H., Bagheri, A., & Yusefi, A. (2017, Spring). Intellectual, psychological, and social capital and business innovation: The moderating effect of organizational culture. *Iranian Journal of Management Studies*, 10(2), 307–333. https://doi.org/10.22059/ijms.2017.215054.672249
- Choi, Y. (2015). The impact of social capital on employees' knowledge-sharing behavior:

An empirical analysis of U.S. federal agencies. *Public Performance & Management Review*, 39(1), 381–405.

https://doi.org/10.1080/15309576.2015.1108795

- Chronéer, D., & Backlund, F. (2015, June/July). A holistic view on learning in project-based organizations. *Project Management Journal*, 46(3), 61–74. https://doi.org/10.1002/pmj.21503
- Chun, J. Y., & Yoong, W. (2015). Social facets of knowledge creation: The validation of knowledge assets. *Social Behavior and Personality*, *43*(5), 815–828. https://doi.org/10.2224/sbp.2015.43.5.815
- Cohen, W. M., & Levinthal, D. A. (1990). Absorptive capacity: A new perspective on learning and innovation. *Administrative Science Quarterly*, *35*(1), 128–152.
- Cole, C. (2017). Project management evolution to improve success in infrastructure projects. *Management Dynamics in the Knowledge Economy*, 5(4), 619–640. <a href="https://doi.org/10.25019/MDKE/5.4.09">https://doi.org/10.25019/MDKE/5.4.09</a>
- Connelly, L. M. (2016). Trustworthiness in qualitative research. *Medsurg Nursing*, 25(6), 435–436.
- Cooper, P. (2016). Informatics: Data, information, knowledge and wisdom. *Anaesthesia*& *Intensive Care Medicine*, 18(1), 55–56.

  <a href="https://doi.org/10.1016/j.mpaic.2016.10.006">https://doi.org/10.1016/j.mpaic.2016.10.006</a>
- Cope, D. G. (2014, January). Methods and meanings: Credibility and trustworthiness of qualitative research. *Oncology Nursing Forum*, 41(1), 89–91.

#### https://doi.org/10.1188/14.ONF.89-91

- Cronin, C. (2014). Using case study research as a rigorous form of inquiry. *Nurse Researcher*, 21(5), 19–27.
- Dădârlat, A., & Dumitrașcu, D. (2015). The role and importance of human capital in contemporary organizations. *Review of Management & Economic Engineering*, 14(1), 78–86.
- Daghfous, A., Belkhodja, O., & Angell, L. C. (2013). Understanding and managing knowledge loss. *Journal of Knowledge Management*, *17*(5), 639–660. http://dx.doi.org/10.1108/JKM-12-2012-0394
- Daher, N. (2016, Winter). The relationships between organizational culture and organizational innovation. *International Journal of Business and Public Administration*, 13(2), 1–15.
- Daniel, B. K. (2019). Using the TACT framework to learn the principles of rigour in qualitative research. *The Electronic Journal of Business Research Methods*, 17(3), 118–129. <a href="https://dx.doi.org/10.34190/JBRM.17.3.002">https://dx.doi.org/10.34190/JBRM.17.3.002</a>
- D'Armagnac, S. (2015, November). Issues in the management of embedded knowledge in project-based organizations: The project actor's role. *Knowledge Management Research & Practice*, 13(4), 446–462. <a href="https://doi.org/10.1057/kmrp.2013.62">https://doi.org/10.1057/kmrp.2013.62</a>
- Dauber, D., Fink, G., & Yolles, M. (2012). A configuration model of organizational culture. *SAGE Open*, 2(1), 1–16. <a href="https://doi.org/10.1177/2158244012441482">https://doi.org/10.1177/2158244012441482</a>
- De Geofroy, Z., & Evans, M. M. (2017). Are emotionally intelligent employees less

- likely to hide their knowledge?. *Knowledge and Process Management*, 24(2), 81–95. https://doi.org/10.1002/kpm.1532
- Demigha, S. (2015, September). *Knowledge management and intellectual capital in an enterprise information system*. Paper presented at the European Conference on Knowledge Management, Udine, Italy.
- De Toni, A. F., Fornasier, A., & Nonino, F. (2017). The nature and value of knowledge. *Kybernetes*, 46(6), 966–979. https://doi.org/10.1108/K-01-2017-0016
- Di Vincenzo, F., & Mascia, D. (2012). Social capital in project-based organizations: Its role, structure, and impact on project performance. *International Journal of Project Management*, 30(1), 5–14. https://doi.org/10.1016/j.ijproman.2011.03.006
- Draper, J. (2015). Ethnography: Principles, practice and potential. *Nursing Standard*, 29(36), 36–41. https://doi.org/10.7748/ns.29.36.36.e8937
- Dumay, J., & Garanina, T. (2013). Intellectual capital research: A critical examination of the third stage. *Journal of Intellectual Capital*, *14*(1), 10–25.

  <a href="https://doi.org/10.1108/14691931311288995">https://doi.org/10.1108/14691931311288995</a></a>
- Dworkin, S. L. (2012). Sample size policy for qualitative studies using in-depth interviews. *Archives of Sexual Behavior*, 41(1), 1319–1320. https://doi.org/10.1007/s10508-012-0016-6
- Echajari, L., & Thomas, C. (2015). Learning from complex and heterogeneous experiences: The role of knowledge codification. *Journal of Knowledge Management*, 19(5), 968-986. https://doi.org/10.1108/JKM-02-2015-0048

- Edvinsson, L., & Malone, M. (1997). *Intellectual capital: Realizing your company's true* value by finding its hidden brainpower. HarperCollins.
- English, C. (2015, October). How to evaluate the quality of new research. *Nursing Children and Young People*, 27(8), 2.
- Ensslin, L., Mussi, C. C., Ensslin, S. R., Dutra, A., & Fontana, L. P. B. (2020). Managing intellectual capital through a collective intelligence approach. *Journal of Intellectual Capital*, *17*(2), 298–319. <a href="http://dx.doi.org/10.1108/JKM-12-2019-0689">http://dx.doi.org/10.1108/JKM-12-2019-0689</a>
- Erden, Z., Klang, D., Sydler, R., & Von Krogh, G. (2014). Knowledge-flows and firm performance. *Journal of Business Research*, 67(1), 2777–2785. https://doi.org/10.1016/j.jbusres.2012.09.001
- Eskerod, P. (1996). Meaning and action in a multi-project environment. Understanding a multi-project environment by means of metaphors and basic assumptions.

  \*International Journal of Project Management, 14(2), 61–65.

  http://dx.doi.org/10.1016/0263-7863(95)00038-0
- Felicio, J. A., Couto, E., & Caiado, J. (2014). Human capital, social capital and organizational performance. *Management Decision*, *52*(2), 350–364. http://dx.doi.org/10.1108/MD-04-2013-0260
- Florczak, K. L. (2017). Adding to the truth of the matter: The case for qualitative research. *Nursing Science Quarterly*, *30*(4), 296–299. https://doi.org/10.1177/0894318417724

- Fong, P. S., & Kwok, C. W. (2009, December). Organizational culture and knowledge management success at project and organizational levels in contracting firms.

  \*Journal of Construction Engineering and Management, 135(12), 1348–1356.\*

  http://dx.doi.org/10.1061/(ASCE)CO.1943-7862.0000106
- Foss, N. J. (2007). The emerging knowledge governance approach: Challenges and characteristics. *Organization*, *14*(1), 29–52. https://doi.org/10.1177/1350508407071859
- Fragouli, E. (2015). Intellectual capital & organizational advantage: An economic approach to its valuation and measurement. *International Journal of Information, Business and Management*, 7(1), 36–57.
- Frels, R. K., & Onwuegbuzie, A. J. (2013). Administering quantitative instruments with qualitative interviews: A mixed research approach. *Journal of Counseling & Development*, 91, 184–194. <a href="https://doi.org/10.1002/j.1556-6676.2013.00085.x">https://doi.org/10.1002/j.1556-6676.2013.00085.x</a>
- Fusch, P. I., & Ness, L. R. (2015). Are we there yet? Data saturation in qualitative research. *The Qualitative Report*, 20(9), 1408–1416.

  www.nova.edu/ssss/QR/QR20/9/fusch1.pdf
- Gaghman, A. (2019). The impact of individual behavioural factors on tacit knowledge retention in oil and gas organisations. *European Journal of Interdisciplinary*Studies, 11(1), 18–28. https://doi.org/10.24818/ejis.2019.02
- Ganescu, C., & Gangone, A. (2017). A model of socially responsible organizational culture. *Studia Universitatis "Vasile Goldis" Arad Economic Sciences*, 27(2),

- 45-59. https://doi.org/10.1515/sues-2017-0008
- García-Fernández, M. (2015). How to measure knowledge management: Dimensions and model. *VINE*, 45(1), 107–125. <a href="https://doi.org/10.1108/VINE-10-2013-0063">https://doi.org/10.1108/VINE-10-2013-0063</a>
- Gaus, N. (2017). Selecting research approaches and research designs: A reflective essay.

  \*Qualitative Research Journal, 17(2), 99–112. <a href="https://doi.org/10.1108/QRJ-07-2016-0041">https://doi.org/10.1108/QRJ-07-2016-0041</a>
- Gelling, L. (2015). Qualitative research. *Nursing Standard*, 29(30), 43–47.
- Gentles, S. J., Charles, C., Ploeg, J., & McKibbon, K. A. (2015). Sampling in qualitative research: Insights from an overview of the methods literature. *The Qualitative Report*, 20(11), 1772–1789.
- Gerritsen, A. L., Stuiver, M., & Termeer, C. (2013). Knowledge governance: An exploration of principles, impact, and barriers. *Science and Public Policy*, 40(5), 604–615. https://doi.org/10.1093/scipol/sct012
- Ghosh, S., Amaya, L., & Skibniewski, M. J. (2012). Identifying areas of knowledge governance for successful projects. *Journal of Civil Engineering and Management*, 18(4), 495–504. https://doi.org/10.3846/13923730.2012.700642
- Gioacasi, D. (2014). Intellectual capital: A critical approach on definitions and categorization. *Centre for European Studies*, 6(4), 57–63.
- Gitinejad, B., & Keramati, M. A. (2013). A study on factors influencing implementation of knowledge management. *Management Science Letters*, *3*(9), 2393–2398. https://doi.org/10.5267/j.msl.2013.08.029

- Goertzen, M. J. (2017, May/June). Introduction to quantitative research and data. *Library Technology Reports*, *53*(4), 12–18. https://journals.ala.org/index.php/ltr/article/view/6325
- Gomezelj, D. O., & Antoncic, B. (2015). Employees' knowledge determinants in SMEs:

  The case of Slovenia. *Journal of Business Economics and Management*, 16(2),
  422–444. https://doi.org/10.3846/16111699.2012.734326
- Grandinetti, R. (2016). Absorptive capacity and knowledge management in small and medium enterprises. *Knowledge Management Research & Practice*, *14*(2), 159–168. https://doi.org/10.1057/kmrp.2016.2
- Grant, R. M. (1996). Toward a knowledge-based theory of the firm. *Strategic Management Journal*, 17, 109–122.
- Guba, E. G., & Lincoln, Y. S. (1994). Competing paradigms in qualitative research. In N. Denzin & Y. Lincoln (Eds.), *Handbook of qualitative research* (pp. 105–117).

  Sage.
- Hadad, S. (2017). Knowledge economy: Characteristics and dimensions. *Management Dynamics in the Knowledge Economy*, 5(2), 203–225. https://doi.org/10.25019/MDKE/5.2.03
- Halcomb, E., & Peters, K. (2016, September 2016). Research would not be possible without participants. *Nurse Researcher*, 24(1), 6–7.
- Hammarberg, K., Kirkman, M., & De Lacey, S. (2016). Qualitative research methods: When to use them and how to judge them. *Human Reproduction*, *31*(3), 498–501.

## https://doi.org/10.1093/humrep/dev334

- Han, J. (2018). Team-bonding and team-bridging social capital: Conceptualization and implications. *Team Performance Management: An International Journal*, 24(1/2), 17–42. <a href="https://doi.org/10.1108/TPM-02-2017-0010">https://doi.org/10.1108/TPM-02-2017-0010</a>
- Hana, U., & Lucie, L. (2011). Staff turnover as a possible threat to knowledge loss. *Journal of Competitiveness*, 3(3), 84–98.
- Handzic, M., Durmic, N., Kraljic, A., & Kraljic, T. (2016). An empirical investigation of the relationship between intellectual capital and project success. *Journal of Intellectual Capital*, 17(3), 471–483. https://doi.org/10.1108/JIC-01-2016-0004
- Harris, L. (2000). A theory of intellectual capital. *Advances in Developing Human Resources*, 2(1), 22–37. https://doi.org/10.1177/152342230000200104
- Hawamdeh, S., & Raigangar, V. (2014). Qualitative interviewing: Methodological challenges in Arab settings. *Nurse Research*, 21(3), 27–31.
- Hislop, D. (2013). *Knowledge management in organizations: A critical introduction* (3rd ed.). Oxford University Press.
- Holten, A., Hancock, G. R., Persson, R., Hansen, Å., & Høgh, A. (2016). Knowledge hoarding: Antecedent or consequent of negative acts? The mediating role of trust and justice. *Journal of Knowledge Management*, 20(2), 215–229.
  <a href="https://doi.org/10.1108/JKM-06-2015-0222">https://doi.org/10.1108/JKM-06-2015-0222</a>
- Houghton, C., Casey, D., Shaw, D., & Murphy, K. (2013, March). Rigour in qualitative case-study research. *Nurse Researcher*, 20(4), 12–17.

- Hsu, I., & Sabherwal, R. (2012, June). Relationship between intellectual capital and knowledge management: An empirical investigation. *Decision Sciences*, 43(3), 489–524.
- Hussinki, H., Ritala, P., Vanhala, M., & Kianto, A. (2017). Intellectual capital, knowledge management practices and firm performance. *Journal of Intellectual Capital*, 18(4), 904–922. https://doi.org/10.1108/JIC-11-2016-0116
- Ienciu, N. M., & Matis, D. (2011). A theoretical framework of intellectual capital.

  \*International Journal of Business Research, 11(2), 131–136. <a href="http://ijbr-journal.org/domains/IJBR-JOURNAL/Default.aspx">http://ijbr-journal.org/domains/IJBR-JOURNAL/Default.aspx</a>
- Ilies, L., & Metz, D. (2017). The link between organizational culture and organizational performance A literature review. *Managerial Challenges of the Contemporary Society*, 10(1), 35–40.
- Indira, V., Suganthi, L., & Anand, S. A. (2012, August). Critical evaluation of knowledge management frameworks for I.T. services organizations. *Advances in Management*, 5(8), 54–65.
- Jaber, O., & Caglar, D. (2017). The role of organizational learning as a mediator in investigating the relationship between knowledge management and organizational performance: The case of banks listed in the stocks exchange of Palestine.
  International Journal of Economic Perspectives, 11(1), 181–198. www.econsociety.org
- Jasimuddin, S. M., & Zhang, Z. (2014). Knowledge management strategy and

- organizational culture. *Journal of the Operational Research Society*, 65(10), 1490–1500. <a href="https://doi.org/10.1057/jors.2013.101">https://doi.org/10.1057/jors.2013.101</a>
- Jayasingam, S., Govindasamy, M., & Singh, S. K. G. (2016). Instilling affective commitment: insights on what makes knowledge workers want to stay.

  \*Management Research Review, 39(3), 266–288. <a href="https://doi.org/10.1108/MRR-03-2014-0060">https://doi.org/10.1108/MRR-03-2014-0060</a>
- Jayawickrama, U., Liu, S., Smith, M. H., Akhtar, P., & Al Bashir, M. M. (2019).
  Knowledge retention in ERP implementations: The context of UK SMEs.
  Production, Planning & Control, 30(10–12), 1032–1047.
  <a href="https://doi.org/10.1080/09537287.2019.1582107">https://doi.org/10.1080/09537287.2019.1582107</a>
- Jennex, M. E. (2014). A proposed method for assessing knowledge loss risk with departing personnel. *VINE Journal of Information and Knowledge Management Systems*, 44(2), 185–209. <a href="https://doi.org/10.1108/VINE-07-2012-0028">https://doi.org/10.1108/VINE-07-2012-0028</a>
- Johansen, M. (2013). The impact of managerial quality on employee turnover. *Public Management Review*, 15(6), 858–877. https://doi.org/10.1080/14719037.2012.725758
- Jørgensen, L. B., Dahl, R., Pedersen, P. U., & Lomborg, K. (2013). Four types of coping with COPD-induced breathlessness in daily living: A grounded theory study.
  Journal of Research in Nursing, 18(6), 520–541.
  <a href="https://doi.org/10.1177/1744987112468443">https://doi.org/10.1177/1744987112468443</a>
- Jugdev, K., & Mathur, G. (2012). Classifying project management resources by

- complexity and leverage. *International Journal of Managing Projects in Business*, 5(1), 105–124. https://doi.org/10.1108/17538371211192928
- Kanade, S. R., Shenoy, V., Prasad, S., & Goyal, K. S. (2015, December). Talent repellents in business processing outsourcing industry. *TSM Business Review*, 3(2), 23–32.
- Karvalics, L. Z., & Dalal, N. (2013, October). Beyond knowledge management: An extended model of knowledge governance. *International Journal of Knowledge Society Research*, 2(4), 59–69. https://doi.org/10.4018/jksr.2011100105
- Khan, S. N. (2014). Qualitative research method: Grounded theory. *International Journal of Business and Management*, 9(11), 224–233.
  <a href="https://doi.org/10.5539/ijbm.v9n11p224">https://doi.org/10.5539/ijbm.v9n11p224</a>
- Khoramin, M., Vakilifard, H., Hosseini, A., & Qasemi, A. (2014). Organizational culture as a key component of intellectual capital. *Kuwait Chapter of Arabian Journal of Business and Management Review*, 3(12a), 175–184.
- Kianto, A., Ritala, P., Spender, J., & Vanhala, M. (2014). The interaction of intellectual capital assets and knowledge management practices in organizational value creation. *Journal of Intellectual Capital*, *15*(3), 362–375.

  <a href="https://doi.org/10.1108/JIC-05-2014-0059">https://doi.org/10.1108/JIC-05-2014-0059</a>
- Kianto, A., Shujahat, M., Hussain, S., Nawaz, F., & Ali, M. (2019). The impact of knowledge management on knowledge worker productivity. *Baltic Journal of Management*, 14(2), 178–197. https://doi.org/10.1108/BJM-12-2017-0404

- Killen, C. P., Jugdev, K., Drouin, N., & Petit, Y. (2012). Advancing project and portfolio management research: Applying strategic management theories. *International Journal of Project Management*, 30(1), 525–538.
  https://doi.org/10.1016/j.ijproman.2011.12.004
- Killingsworth, B., Xue, Y., & Liu, Y. (2016). Factors influencing knowledge sharing among global virtual teams. *Team Performance Management*, 22(5/6), 284–300. https://doi.org/10.1108/TPM-10-2015-0042
- Kim, T. T., Lee, G., Paek, S., & Lee, S. (2013). Social capital, knowledge sharing and organizational performance: What structural relationship do they have in hotels?.
  International Journal of Contemporary Hospitality Management, 25(5), 683–704.
  <a href="https://doi.org/10.1108/IJCHM-Jan-2012-0010">https://doi.org/10.1108/IJCHM-Jan-2012-0010</a>
- Koohang, A., Paliszkiewicz, J., & Goluchowski, J. (2017). The impact of leadership on trust, knowledge management and organizational performance: A research model.
  Industrial Management & Data Systems, 117(3), 521–537.
  <a href="https://doi.org/10.1108/IMDS-02-2016-0072">https://doi.org/10.1108/IMDS-02-2016-0072</a>
- Koskinen, K. U. (2010). Organizational memories in project-based companies: An autopoietic view. *The Learning Organization*, *17*(2), 149–162. https://doi.org/10.1108/09696471011019862
- Koskinen, K. U. (2013). Business organizations' knowledge-production processes: An autopoietic approach. *International Journal of Organizational Analysis*, 21(2), 137–153. https://doi.org/10.1108/IJOA-05-2011-0490

- Kruth, J. G. (2015, Fall). Five qualitative research approaches and their applications in parapsychology. *The Journal of Parapsychology*, 79(2), 219–233.
- Kuruppuge, R. H., & Gregar, A. (2018). Employee's learning in the organization: A study of knowledge-based industries. *Foundations of Management*, *10*(1), 13–22. https://doi.org/10.2478/fman-2018-0002
- Langtree, T., Birks, M., & Biedermann, N. (2019). Separating "fact" from fiction:

  Strategies to improve rigour in historical research. *Forum: Qualitative Social*Research, 20(2). https://dx.doi.org/10.17169/fqs-20.2.3196
- Lee, J., Park, J., & Lee, S. (2015a). Raising team social capital with knowledge and communication in information systems development projects. *International Journal of Project Management*, 33(1), 797–807.

  <a href="https://doi.org/10.1016/j.ijproman.2014.12.001">https://doi.org/10.1016/j.ijproman.2014.12.001</a>
- Lee, S., Fernandez, S., & Chang, C. (2018). Job scarcity and voluntary turnover in the U.S. federal bureaucracy. *Public Personnel Management*, 47(1), 3–25. https://doi.org/10.1177/0091026017732798
- Lee, S., Park, J., & Lee, J. (2015b). Explaining knowledge sharing with social capital theory in information systems development projects. *Industrial Management & Data Systems*, 115(5), 883–900. https://doi.org/10.1108/IMDS-01-2015-0017
- Lee, S., Yoo, Y., & Yun, S. (2015). Sharing my knowledge? An interactional perspective.

  \*\*Journal of Managerial Psychology, 30(8), 986–1002.\*\*

  https://doi.org/10.1108/JMP-11-2013-0355

- Lemon, L. L., & Hayes, J. (2020). Enhancing trustworthiness of qualitative findings:

  Using Leximancer for qualitative data analysis triangulation. *The Qualitative*Report, 25(3), 604–614. https://nsuworks.nova.edu/tqr/vol25/iss3/3
- Levallet, N., & Chan, Y. E. (2019, April). Organizational knowledge retention and knowledge loss. *Journal of Knowledge Management*, 23(1), 176–199. http://dx.doi.org/10.1108/JKM-08-2017-0358
- Levy, M. (2011). Knowledge retention: Minimizing organizational business loss. *Journal of Knowledge Management*, *15*(4), 582–600. http://dx.doi.org/10.1108/13673271111151974
- Lin, T., Chang, C. L., & Tsai, W. (2016). The influences of knowledge loss and knowledge retention mechanisms on the absorptive capacity and performance of a MIS department. *Management Decision*, *54*(7), 1757–1787.

  https://doi.org/10.1108/MD-02-2016-0117
- Little, T. A., & Deokar, A. V. (2016). Understanding knowledge creation in the context of knowledge-intensive business processes. *Journal of Knowledge Management*, 25(5), 858–879. https://doi.org/10.1108/JKM-11-2015-0443
- Liu, M., Lin, C., Chen, M., Chen, P., & Chen, K. (2020). Strengthening knowledge sharing and job dedication: The roles of corporate social responsibility and ethical leadership. *Leadership & Organization Development Journal*, 41(1), 73–87. <a href="https://doi.org/10.1108/LODJ-06-2019-0278">https://doi.org/10.1108/LODJ-06-2019-0278</a>
- Lo, J. (2015). The information technology workforce: A review and assessment of

- voluntary turnover research. *Information Systems Frontiers*, 22(7), 1573–1590. https://doi.org/10.1007/s10796-013-9408-y
- Lopez, V. W., & Esteves, J. (2013). Acquiring external knowledge to avoid wheel reinvention. *Journal of Knowledge Management*, *17*(1), 87–105. https://doi.org/10.1108/13673271311300787
- Lyons, E., & Coyle, A. (Eds.). (2016). Analyzing qualitative data in psychology. Sage.
- Ma, Z., Huang, Y., Wu, J., Dong, W., & Qi, L. (2014). What matters for knowledge sharing in collectivistic cultures? Empirical evidence from China. *Journal of Knowledge Management*, *18*(5), 1004–1019. <a href="https://dx.doi.org/10.1108/JKM-06-2014-0252">https://dx.doi.org/10.1108/JKM-06-2014-0252</a>
- Mageswari, S. U., Sivasubramanian, C., & Dath, T. S. (2016). The impact of government initiatives on knowledge management processes: An empirical analysis. *The IUP Journal of Knowledge Management*, *14*(1), 7–32.
- Mainga, W. (2017). Examining project learning, project management competencies, and project efficiency in project-based firms (PBFs). *International Journal of Managing Projects in Business*, 10(3), 454–504. <a href="https://doi.org/10.1108/IJMPB-04-2016-0035">https://doi.org/10.1108/IJMPB-04-2016-0035</a>
- Makhubela, S., & Ngoepe, M. (2018). Knowledge retention in a platinum mine in the north west province of South Africa. *South African Journal of Information Management*, 14(1), 7–32. <a href="https://doi.org/10.4102/sajim.v20i1.905">https://doi.org/10.4102/sajim.v20i1.905</a>
- Manuti, A., Impedovo, M. A., & De Palma, P. D. (2017). Managing social and human

- capital in organizations. *Journal of Workplace Learning*, 29(3), 217–234. http://dx.doi.org/10.1108/JWL-07-2016-0062
- Mariano, S., & Walter, C. (2015). The construct of absorptive capacity in knowledge management and intellectual capital research: Content and text analyses. *Journal of Knowledge Management*, 19(2), 372–400. <a href="https://doi.org/10.1108/JKM-08-2014-0342">https://doi.org/10.1108/JKM-08-2014-0342</a>
- Marsden, T. (2016). What is the true cost of attrition? *Strategic HR Review*, 15(4), 189–190.
- Marshall, B., Cardon, P., Poddar, A., & Fontenot, R. (2013, Fall). Does sample size matter in qualitative research? A review of qualitative interviews in IS research. *The Journal of Computer Information Systems*, 54(1), 11–22.
- Martelo-Landroguez, S., & Cepeda-Carrión, G. (2016). How knowledge management processes can create and capture value for firms? *Knowledge Management Research & Practice*, *14*(4), 423–433. https://doi.org/10.1057/kmrp.2015.26
- Martins, E. C., & Meyer, H. W. (2012). Organizational and behavioral factors that influence knowledge retention. *Journal of Knowledge Management*, *16*(1), 77–96. http://dx.doi.org/10.1108/13673271211198954
- Mason, M. (2010). Sample size and saturation in PhD studies using qualitative interviews. *Forum: Qualitative Social Research*, 11(3). https://www.qualitativeresearch.net/index.php/fqs/article/view/1428
- Massingham, P. (2014). An evaluation of knowledge management tools: Part 1 –

- managing knowledge resources. *Journal of Knowledge Management*, 18(6), 1076–1100. https://doi.org/10.1108/JKM-11-2013-0449
- Massingham, P. R. (2008). Measuring the impact of knowledge loss: More than ripples on a pond. *Management Learning*, *39*(5), 541–560. https://doi.org/10.1177/1350507608096040
- Massingham, P. R. (2018). Measuring the impact of knowledge loss: A longitudinal study. *Journal of Knowledge Management*, 22(4), 721–758. https://doi.org/10.1108/JKM-08-2016-0338
- Massingham, P. R., & Massingham, R. K. (2014). Does knowledge management produce practical outcomes? *Journal of Knowledge Management*, 18(2), 221–254. http://dx.doi.org/10.1108/JKM-10-2013-0390
- Massingham, P. R., & Tam, L. (2015). The relationship between human capital, value creation and employee reward. *Journal of Intellectual Capital*, *16*(2), 390–418. http://dx.doi.org/10.1108/JIC-06-2014-0075
- Matua, G. A. (2015, March). Choosing phenomenology as a guiding philosophy for nursing research. *Nurse Researcher*, 22(4), 30–34.
  <a href="https://doi.org/10.7748/nr.22.4.30.e1325">https://doi.org/10.7748/nr.22.4.30.e1325</a>
- Maxwell, J. A. (2013). *Qualitative research design: An interactive approach* (3rd ed.). SAGE Publications.
- Mayan, M. J. (2016). Essentials of qualitative inquiry. Routledge.
- Micic, R. (2015). Leadership role in certain phases of knowledge management processes.

- Ekonomika, 61(4), 47–56. <a href="https://doi.org/10.5937/ekonomika1504047M">https://doi.org/10.5937/ekonomika1504047M</a>
- Miles, M. B., & Huberman, A. M. (1994). *Qualitative Data Analysis: An expanded sourcebook* (2nd ed.). Sage Publications.
- Millar, C. C., Lockett, M., & Mahon, J. F. (2016). Knowledge intensive organisations:

  On the frontiers of knowledge management. *Journal of Knowledge Management*,

  20(5), 845–857. https://doi.org/10.1108/JKM-07-2016-0296
- Miterev, M., Turner, J. R., & Mancini, M. (2017). The organization design perspective on the project-based organization: A structured review. *International Journal of Managing Projects in Business*, 10(3), 527–549. <a href="https://doi.org/10.1108/IJMPB-06-2016-0048">https://doi.org/10.1108/IJMPB-06-2016-0048</a>
- Mitrevski, V., & Aceski, A. (2017, January). Management of knowledge in organizations with different activities. *Indian Journal of Commerce & Management Studies*, 8(1), 22–27. <a href="https://doi.org/10.18843/ijcms/v8i1/03">https://doi.org/10.18843/ijcms/v8i1/03</a>
- Mohajan, H. K. (2018). Qualitative research methodology in social sciences and related subjects. *Journal of Economic Development, Environment and People*, 7(1), 23–48.
- Molodchik, M., & Jardon, C. (2015). Facilitating organizational learning in the Russian business context. *The Learning Organization*, 22(6), 306–316. https://doi.org/10.1108/TLO-11-2014-0061
- Morse, J. M. (2015). Critical analysis of strategies for determining rigor in qualitative inquiry. *Qualitative Health Research*, 25(9), 1212–1222.

#### https://doi.org/10.1177/1049732315588501

- Motshegwa, N. (2017). Knowledge retention strategies employed by Botswana's tourism and hospitality industry: The case of Gaborone. *Tourism Management Perspectives*, 24(1), 107–110. https://doi.org/10.1016/j.tmp.2017.07.022
- Moud, H. I., & Abbasnejad, B. (2012). Factors affecting knowledge transfer in project-based organizations. Paper presented at the 11th International Conference of the Asia Chapter & 2nd Conference of the MENA Chapter of the Academy of Human Resource Development & 10th International Conference on Knowledge, Economy and Management, Istanbul, TK.
- Mousavizadeh, M., Harden, G., Ryan, S., & Windsor, J. (2015, Summer). Knowledge management and the creation of business value. *The Journal of Computer Information Systems*, 55(4), 35–45.
- Munn, Z., Porritt, K., Lockwood, C., Aromataris, E., & Pearson, A. (2014). Establishing confidence in the output of qualitative research synthesis: The ConQual approach.

  \*\*BMC Medical Research Methodology, 14, 1–7. <a href="https://doi.org/10.1186/1471-2288-14-108">https://doi.org/10.1186/1471-2288-14-108</a>
- Munoz, C. A., Mosey, S., & Binks, M. (2015). The tacit mystery: Reconciling different approaches to tacit knowledge. *Knowledge Management Research & Practice*, 13(3), 289–298. http://dx.doi.org/10.1057/kmrp.2013.50
- Mura, M., Lettieri, E., Radaelli, G., & Spiller, N. (2016). Behavioural operations in healthcare: A knowledge sharing perspective. *International Journal of Operations*

- & Production Management, 36(10), 1222–1246. https://dx.doi.org/10.1108/IJOPM-04-2015-0234
- Nafukho, F. M. (2009, June). HRD's role in identifying, measuring, and managing knowledge assets in the intangible economy. *Advances in Developing Human Resources*, 11(3), 399–410. http://dx.doi.org/10.1177/1523422309337718
- Nazari, J. A., Herremans, I. M., Isaac, R. G., Manassian, A., & Kline, T. J. (2011).
  Organizational culture, climate and intellectual capital: An interaction analysis.
  Journal of Intellectual Capital, 12(2), 224–248.
  <a href="https://doi.org/10.1108/14691931111123403">https://doi.org/10.1108/14691931111123403</a>
- Nhon, H. T., Thong, B. Q., & Phuong, N. V. (2018). The impact of intellectual capital dimensions on Vietnamese information communication technology firm performance: A mediation analysis of human and social capital. *Academy of Strategic Management Journal*, 17(1), 1–15.
- Nonaka, I., & Takeuchi, H. (1995). *The knowledge creating company*. Oxford University Press.
- Novak, R., Roblek, V., & Devetak, G. (2013, May-June). Relation between knowledge management and turnover in Slovenian micro and small start-up organizations.

  Organizacija, 46(3), 99–107. https://doi.org/10.2478/orga-2013-0008
- O'Brien, M. A., Carson, A., Barbera, L., Brouwers, M. C., Earle, C. C., Graham, I. D., Mittmann, N., & Grunfeld, E. (2018). Variable participation of knowledge users in cancer health services research: results of a multiple case study. *BMC Medical*

- Research Methodology, 18, 1–13. https://doi.org/10.1186/s12874-018-0593-8
- O'Donoghue, N., & Croasdell, D. T. (2009). Protecting knowledge assets in multinational enterprises: A comparative case approach. *VINE: The journal of information and knowledge management systems*, *39*(4), 298–319.

  https://doi.org/10.1108/03055720911013616
- Odor, H. O. (2019). A literature review on organizational learning and learning organizations. *International Journal of Information, Business and Management*, 11(3), 281–295.
- Onwuegbuzie, A. J., & Byers, V. T. (2014). An exemplar for combining the collection, analysis, and interpretations of verbal and nonverbal data in qualitative research.

  \*International Journal of Education, 6(5), 533–544.\*

  https://dx.doi.org/10.5296/ije.v6il.4399
- Oun, T. A., Blackburn, T. D., Olson, B. A., & Blessner, P. (2016, September). An enterprise-wide knowledge management approach to project management. *Engineering Management Journal*, 28(3), 176–192.
- Palinkas, L. A., Horowitz, S. M., Green, C. A., Wisdom, J. P., & Duan, N. (2015, September). Purposeful sampling for qualitative data collection and analysis in mixed method implementation research. *Administration and Policy in Mental Health and Mental Health Services Research*, 42(5), 533–544.

  <a href="http://dx.doi.org/10.1007/s10488-013-0528-y">http://dx.doi.org/10.1007/s10488-013-0528-y</a>
- Parboteeah, P., Jackson, T., & Wilkinson, N. (2016). A theoretically grounded model to

- reduce the risk of knowledge loss in organizations: An energy company evaluation. *Knowledge and Process Management*, 23(3), 171–183. https://doi.org/10.1002/kpm.1502
- Park, J., & Park, M. (2016). Qualitative versus quantitative research methods: Discovery or justification? *Journal of Marketing Thought*, *3*(1), 1–7. https://doi.org/10.15577/jmt.2016.03.01.1
- Park, S., & Eun-Jee, K. (2018). Fostering organizational learning through leadership and knowledge sharing. *Journal of Knowledge Management*, 22(6), 1408–1423. https://dx.doi.org/10.1108/JKM-10-2017-0467
- Park, S., & Kim, E. (2015). Revisiting knowledge sharing from the organizational change perspective. *European Journal of Training and Development*, *39*(9), 769–797. https://doi.org/10.1108/EJTD-06-2015-0042
- Patriotta, G., Castellano, A., & Wright, M. (2013). Coordinating knowledge transfer:

  Global managers as higher-level intermediaries. *Journal of World Business*, 48(4),

  515–526. <a href="https://dx.doi.org/10.1016/j.jwb.2012.09.007">https://dx.doi.org/10.1016/j.jwb.2012.09.007</a>
- Patton, M. Q. (2015). *Qualitative research & evaluation methods* (4th ed.). Sage Publications.
- Pee, L. G., Kankanhalli, A., Tan, G. W., & Tham, G. Z. (2014). Mitigating the impact of member turnover in information systems development projects. *IEEE Transactions on Engineering Management*, 61(4), 702–716.
  <a href="https://dx.doi.org/10.1109/TEM.2014.2332339">https://dx.doi.org/10.1109/TEM.2014.2332339</a>

- Pellegrini, M. M., Ciampi, F., Marzi, G., & Orlando, B. (2020). The relationship between knowledge management and leadership: Mapping the field and providing future research avenues. *Journal of Knowledge Management*, 24(6), 1445–1492. https://dx.doi.org/10.1108/JKM-01-2020-0034
- Pemsel, S., & Müller, R. (2012, November). The governance of knowledge in project-based organizations. *International Journal of Project Management*, *30*(8), 865–876. https://dx.doi.org/10.1016/j.ijproman.2012.02.002
- Pemsel, S., Müller, R., & Söderlund, J. (2016). Knowledge governance strategies in project-based organizations. *Long Range Planning*, 49(1), 648–660. https://dx.doi.org/10.1016/j.lrp.2016.01.001
- Pemsel, S., Wiewiora, A., Müller, R., Aubry, M., & Brown, K. (2014, November). A conceptualization of knowledge governance in project-based organizations.

  \*International Journal of Project Management, 32(8), 1411–1422.

  http://dx.doi.org/dx.doi.org/10.1016/j.ijproman.2014.01.010
- Percy, W. H., Kostere, K., & Kostere, S. (2015). Generic qualitative research in psychology. *The Qualitative Report*, 20(2), 76–85.

  <a href="http://www.nova.edu/ssss/QR/QR20/2/percy5.pdf">http://www.nova.edu/ssss/QR/QR20/2/percy5.pdf</a>
- Peter, E. (2015). The ethics in qualitative health research: Special considerations. *Ciencia & Saude Coletiva*, 20(9), 2625–2630. <a href="https://dx.doi.org/10.1590/1413-81232015209.06762015">https://dx.doi.org/10.1590/1413-81232015209.06762015</a>
- Polanyi, M. (1966). The tacit dimension (1st ed.). Doubleday.

- Qi, C., & Chau, P. Y. (2018). Will enterprise social networking systems promote knowledge management and organizational learning? An empirical study. *Journal of Organizational Computing and Electronic Commerce*, 28(1), 31–57. https://dx.doi.org/10.1080/10919392.2018.1407081
- Radenkovic, S. D., Devedzic, V., Jovanovic, J., & Jeremic, Z. (2014). Content and knowledge provision service a way to build intellectual capital in learning organizations. *Knowledge Management Research & Practice*, *12*(3), 297–309. <a href="http://dx.doi.org/10.1057/kmrp.2013.55">http://dx.doi.org/10.1057/kmrp.2013.55</a>
- Rajhans, K. (2018). Effective communication management: A key to stakeholder relationship management in project-based organizations. *IUP Journal of Soft Skills*, 12(4), 47–66.
- Ramadan, B. M., Dahiyat, S. E., Bontis, N., & Al-dalahmeh, M. A. (2017). Intellectual capital, knowledge management and social capital within the ICT sector in Jordan. *Journal of Intellectual Capital*, *18*(2), 437–462.

  <a href="https://doi.org/10.1108/JIC-06-2016-0067">https://doi.org/10.1108/JIC-06-2016-0067</a>
- Ramona, T., & Alexandra, B. (2019). Knowledge retention within small and medium-sized enterprises. *Studies in Business and Economics*, *14*(3), 231–238. https://doi.org/10/2478/sbe-2019-0056
- Randeree, E. (2006). Knowledge management: Securing the future. *Journal of Knowledge Management*, 10(4), 145–156.

  <a href="https://doi.org/10.1108/13673270610679435">https://doi.org/10.1108/13673270610679435</a>

- Rashid, M., Clarke, P. M., & O'Connor, R. V. (2019). A systematic examination of knowledge loss in open source software projects. *International Journal of Information Management*, 46(1), 104–123. https://doi.org/10.1016/j.ijinfomgt.2018.11.015
- Raudeliūnienė, J., Davidavičienė, V., & Petrusevičius, R. (2018). A systematic examination of knowledge loss in open source software projects. *International Journal of Information Management*, 46(1), 104–123. <a href="https://doi.org/10.1016/j.ijinfomgt.2018.11.015">https://doi.org/10.1016/j.ijinfomgt.2018.11.015</a>
- Reich, B. H., Gemino, A., & Sauer, C. (2012, May). Knowledge management and project-based knowledge in it projects: A model and preliminary empirical results. *International Journal of Project Management*, 30(3), 663–674. https://doi.org/10.1016/j.ijproman.2011.12.003
- Reich, B. H., Gemino, A., & Sauer, C. (2014, May). How knowledge management impacts performance in projects: An empirical study. *International Journal of Project Management*, 32(4), 590–602. https://doi.org/10.1016/j.ijproman.2013.09.004
- Reiter, S., Stewart, G., & Bruce, C. (2011). A strategy for delayed research method selection: Deciding between grounded theory and phenomenology. *The Electronic Journal of Business Research Methods*, 9(1), 35–46.
- Ren, X., Deng, X., & Liang, L. (2018). Knowledge transfer between projects within project-based organizations: the project nature perspective. *Journal of Knowledge*

- Management, 22(5), 1082–1103. https://doi.org/10.1108/JKM-05-2017-0184
- Rezaei-Zadeh, M., & Darwish, T. K. (2016). Antecedents of absorptive capacity: A new model for developing learning processes. *The Learning Organization*, 23(1), 77–91. <a href="https://doi.org/10.1108/TLO-04-2015-0026">https://doi.org/10.1108/TLO-04-2015-0026</a>
- Riaz, M. N., & Khalili, M. T. (2014). Transformational, transactional leadership and rational decision making in services providing organizations: Moderating role of knowledge management processes. *Pakistan Journal of Commerce and Social Sciences*, 8(2), 355–364.
- Robinson, O. C. (2014). Sampling in interview-based qualitative research: A theoretical and practical guide. *Qualitative Research in Psychology*, 11(1), 25–41. https://doi.org/10.1080/14780887.2013.801543
- Ross, K. M. (2013). Purposeful mentoring in academic libraries. *Journal of Library Administration*, 53(7-8), 412–428. https://doi.org/10.1080/01930826.2013.882195
- Rossetto, K. R. (2014). Qualitative research interviews: Assessing the therapeutic value and challenges. *Journal of Social and Personal Relationships*, *31*(4), 482–489. https://doi.org/10.1177/0265407514522892
- Roulston, K. (2016). Issues involved in methodological analyses of research interviews.

  \*Qualitative Research Journal\*, 16(1), 68–79. <a href="https://doi.org/10.1108/QRJ-02-2015-0015">https://doi.org/10.1108/QRJ-02-2015-0015</a>
- Sabir, H. M., & Kalyar, M. N. (2013, January). Firm's innovativeness and employee job satisfaction: The role of an organizational learning culture. *Interdisciplinary*

- Journal of Contemporary Research in Business, 4(9), 670–686. ijcrb.webs.com
- Saldana, J. (2009). *The coding manual for qualitative researchers* (3rd ed.). Sage Publications.
- Salmaninezhad, R. A., & Daneshvar, M. (2012, February). Relationship analysis between intellectual capital and knowledge management (Case study: Tehran Science & Technology Park). *Interdisciplinary Journal of Contemporary Research in Business*, *3*(10), 135–143. http://ijcrb.webs.com
- Sanchez-Canizares, S. M., Munoz, M. A., & Lopez-Guzman, T. (2007). Organizational culture and intellectual capital: A new model. *Journal of Intellectual Capital*, 8(3), 409–430. <a href="https://doi.org/10.1108/14691930710774849">https://doi.org/10.1108/14691930710774849</a>
- Sanz, M. M. M., & Ortiz-Marcos, I. (2020). Dimensions of knowledge governance in an multi-PMO project context. *International Journal of Managing Projects in Business*, *13*(7), 1423–1441. https://doi.org/10.1108/IJMPB-11-2018-0244
- Sathishkumar, A. S., & Karthikeyan, P. (2017, April). A study on effective organizational learning through knowledge management model. *International Journal of Research in Commerce & Management*, 8(4), 34–36. http://ijrcm.org.in/
- Sayadi, S., Saljughi, Z. S., & Bahraminejad, Z. (2013, July). Intellectual capital and organizational entrepreneurship. *Interdisciplinary Journal of Contemporary Research in Business*, *5*(3), 612–626. http://ijcrb.webs.com
- Schacht, S., Morana, S., & Maedche, A. (2015). The evolution of design principles enabling knowledge reuse for projects: An action design research project. *Journal*

- of Information Technology Theory and Application, 16(3), 5–36.
- Schmitt, A., Borzillo, S., & Probst, G. (2011). Don't let knowledge walk away:

  Knowledge retention during employee downsizing. *Management Learning*, 43(1),
  53–74. http://dx.doi.org/10.1177/1350507611411630
- Secundo, G., Dumay, J., Schiuma, G., & Passiante, G. (2016). Managing intellectual capital through a collective intelligence approach. *Journal of Intellectual Capital*, 17(2), 298–319. http://dx.doi.org/10.1108/JIC-05-2015-0046
- Segundo, R. S. (2002). A new concept of knowledge. *Online Information Review*, 26(4), 239–245. https://doi.org/10.1108/14684520210438688
- Seleim, A. A., & Khalil, O. E. (2011). Understanding the knowledge management-intellectual capital relationship: A two-way analysis. *Journal of Intellectual Capital*, 12(4), 586–614. http://dx.doi.org/10.1108/14691931111181742
- Serra, A. (2016). Architectural documentation through thick description. *Enquiry: The ARCC Journal of Architectural Research*, *13*(2), 17–29. http://dx.doi.org/10.17831/enq:arcc.v13i2.400
- Shannon-Baker, P. (2016). Making paradigms meaningful in mixed methods research.

  \*Journal of Mixed Methods Research, 10(4), 319–334.

  http://dx.doi.org/10.1177/1558689815575861
- Showry, M., & Sayulu, K. (2017). The impact of core self-evaluation on attrition in IT industry. *The IUP Journal of Soft Skills*, 11(4), 7–22.
- Simon, M. K. (2013). Dissertation and scholarly research recipes for success (2nd ed.).

- CreateSpace Independent Publishing.
- Singh, R. M., & Gupta, M. (2014). Knowledge management in teams: Empirical integration and development of a scale. *Journal of Knowledge Management*, 18(4), 777–794. https://doi.org/10.1108/JKM-11-2013-0450
- Singh, S. K. (2008). Role of leadership in knowledge management: A study. *Journal of Knowledge Management*, 12(4), 3–15. https://doi.org/10.1108/13673270810884219
- Sinkovics, R. R., & Alfoldi, E. A. (2012). Progressive focusing and trustworthiness in qualitative research. *Management International Review*, 52(6), 817–845. https://doi.org/10.1007/s11575-012-0140-5
- Sirorei, E. C., & Fombad, M. C. (2019). Knowledge management processes at St Paul's University Library in Kenya. *South African Journal of Information Management*, 21(1). <a href="https://doi.org/10.4102/sajim.v21i1.946">https://doi.org/10.4102/sajim.v21i1.946</a>
- Sitlington, H., & Marshall, V. (2011). Do downsizing decisions affect organizational knowledge and performance?. *Management Decision*, 49(1), 116–129. https://doi.org/10.1108/00251741111094473
- Smith, M. E. (2003). Another road to evaluating knowledge assets. *Human Resource Development Review*, 2(1), 6–25. <a href="https://doi.org/10.1177/1534484302250598">https://doi.org/10.1177/1534484302250598</a>
- Spradley, J. P. (2016). *Participant Observation*. Waveland Press.
- Stulgiene, A., & Ciutiene, R. (2012). HRM challenges in transition to project management (project-based organization). *Economics and Management*, 17(3),

- 1214–1218. http://dx.doi.org/dx.doi.org/10.5755/j01.em.17.3.2145
- Sumbal, M. S., Tsui, E., Cheong, R., & See-to, E. W. (2018). Critical areas of knowledge loss when employees leave in the oil and gas industry. *Journal of Knowledge Management*, 22(7), 1573–1590. <a href="https://doi.org/10.1108/JKM-08-2017-0373">https://doi.org/10.1108/JKM-08-2017-0373</a>
- Supartha, W. G., & Kumala Ratih, I. A. (2017, July). Antecedents of absorptive capacity:

  A proof of proposition. *Journal of Business and Retail Management Research*,

  11(4), 90–107.
- Swart, J., & Kinnie, N. (2013, April). Managing multidimensional knowledge assets: HR configurations in professional service firms. *Human Resource Management Journal*, 23(2), 160–179. http://dx.doi.org/10.1111/j.1748-8583.2012.00197.x
- Tang, V., Yanine, F., & Valenzuela, L. (2016). Data, information, knowledge and intelligence: The mega-nano hypothesis and its implications in innovation.
  International Journal of Innovation Science, 8(3), 199–216.
  https://doi.org/10.1108/IJIS-07-2016-0022
- Tarride, M. I., & Osorio-Vega, P. (2013). Complexity and intellectual capital in organizations. *Kybernetes*, 42(4), 544–553. <a href="https://doi.org/10.1108/K-08-2012-0015">https://doi.org/10.1108/K-08-2012-0015</a>
- Taylor, R., & Thomas-Gregory, A. (2015). Case study research. *Nursing Standard*, 29(41), 36–40. <a href="https://doi.org/10.7748/ns.29.41.36.e8856">https://doi.org/10.7748/ns.29.41.36.e8856</a>
- Terzieva, M., & Morabito, V. (2016). Learning from experience: The project team is the key. *Business Systems Research*, 7(1), 1–15. <a href="https://doi.org/10.1515/bsrj-2016-">https://doi.org/10.1515/bsrj-2016-</a>

- Thomas, E., & Magilvy, J. K. (2011). Qualitative rigor or research validity in qualitative research. *Journal for Specialist Pediatric Nursing*, *16*, 151–155. https://doi.org/10.1111/j.1744-6155.2011.00283.x
- Tissayakorn, K., Akagi, F., & Song, Y. (2013, November). A model of organization knowledge management maturity. *International Journal of Bioscience*, *Biochemistry and Bioinformatics*, *3*(6), 614–618. https://doi.org/10.7763/IJBBB.2013.V3.287
- Tongo, C. I. (2013, June). A performance model for knowledge-based firms: Lessons for managers. *International Journal of Management*, 30(2), 704–716. https://search.proquest.com/openview/3d86b805208129a971b9255f17ae514d/1?pq-origsite=gscholar&cbl=5703
- Turner, J. R., Zimmerman, T., & Allen, J. M. (2012). Teams as a sub-process for knowledge management. *Journal of Knowledge Management*, *16*(6), 963–977. https://doi.org/10.1108/13673271211276227
- Turner, N., Maylor, H., & Swart, J. (2015). Ambidexterity in projects: An intellectual capital perspective. *International Journal of Project Management*, *33*(1), 177–188. <a href="https://doi.org/10.1016/j.ijproman.2014.05.002">https://doi.org/10.1016/j.ijproman.2014.05.002</a>
- Uma Devi, Y., & Murali Krishna, V. (2016, October). Impact of human resources management practices on employee turnover An empirical study with special reference to IT sector. *Journal of Strategic Human Resource Management*, 5(3),

- 50-54.
- Umasankar, M., & Ashok, J. (2013, March). A study on linkage between organizational practices and employee attrition. *International Journal of Knowledge*Management & Practices, 1(1), 30–34.
- Vale, J., Branco, M. C., & Ribeiro, J. (2016). Individual intellectual capital versus collective intellectual capital in a meta-organization. *Journal of Intellectual Capital*, 17(2), 279–297. https://doi.org/10.1108/JIC-05-2015-0044
- Van Manen, M. (2016). Phenomenology of practice: Meaning-giving methods in phenomenological research and writing. Routledge.
- Vidotto, J. D., Ferenhof, H. A., Selig, P. M., & Bastos, R. C. (2017). A human capital measurement scale. *Journal of Intellectual Capital*, 18(2), 316–329. https://doi.org/10.1108/JIC-08-2016-0085
- Vo, L. (2012). Pragmatist perspective on knowledge and knowledge management in organizations. *International Business Research*, *5*(9), 78–88. <a href="https://doi.org/10.5539/ibr.v5n9p78">https://doi.org/10.5539/ibr.v5n9p78</a>
- Wang, Z., Wang, N., Cao, J., & Ye, X. (2016). The impact of intellectual capital knowledge management strategy fit on firm performance. *Management Decision*, 54(8), 1861–1885. <a href="https://doi.org/10.1108/MD-06-2015-0231">https://doi.org/10.1108/MD-06-2015-0231</a>
- Wang, Z., Wang, N., & Liang, H. (2014). Knowledge sharing, intellectual capital and firm performance. *Management Decision*, 52(2), 230–258. <a href="http://dx.doi.org/10.1108/MD-02-2013-0064">http://dx.doi.org/10.1108/MD-02-2013-0064</a>

- Watson, F. A. (2016). Lessons learned on approaches to data collection and analysis from a pilot study. *Nurse Researcher*, 24(1), 32–36. https://doi.org/10.7748/nr.2016.e1444
- Watson, R. (2015). Quantitative Research. *Nursing Standard* (2014+), 29(31), 44. http://dx.doi.org/10.7748/ns.29.31.44.e8681
- Whelan, E., & Carcary, M. (2011). Integrating talent and knowledge management: Where are the benefits? *Journal of Knowledge Management*, 15(4), 675–687. http://dx.doi.org/10.1108/13673271111152018
- Widjaja, J. H., & Kuslina, B. H. (2018). The role of organizational culture and knowledge management to encourage innovation in governance in an Indonesian private university. *Review of Integrative Business and Economics Research*, (1), 255–265.
- Wikström, E., Eriksson, E., Karamehmedovic, L., & Liff, R. (2018). Knowledge retention and age management senior employees' experiences in a Swedish multinational company. *Journal of Knowledge Management*, 22(7), 1510–1526. <a href="https://doi.org/10.1108/JKM-09-2017-0442">https://doi.org/10.1108/JKM-09-2017-0442</a>
- Wray, J., Archibong, U., & Walton, S. (2014). Why undertake a pilot in a qualitative PhD study? Lessons learned to promote success. *Nurse Researcher*, 24(3), 31–39. https://doi.org/10.7748/nr.2017.e1416
- Wu, I., & Chen, J. (2014). Knowledge management driven firm performance: The roles of business process capabilities and organizational learning. *Journal of*

- *Knowledge Management*, 18(6), 1141–1164. <a href="http://dx.doi.org/10.1108/JKM-05-2014-0192">http://dx.doi.org/10.1108/JKM-05-2014-0192</a>
- Yaghootkar, K., & Gil, N. (2012). The effects of schedule-driven project management in multi-project environments. *International Journal of Project Management*, 30(1), 127–140. <a href="http://dx.doi.org/10.1016/j.ijproman.2011.02.005">http://dx.doi.org/10.1016/j.ijproman.2011.02.005</a>
- Yang, L., Huang, C., & Hsu, T. (2014). Knowledge leadership to improve project and organizational performance. *International Journal of Project Management*, 32(1), 40–53. https://doi.org/10.1016/j.ijproman.2013.01.011
- Yazici, H. J. (2011). Significance of organizational culture in perceived project and business performance. *Engineering Management Journal*, 23(2), 20–29. https://doi.org/10.1080/10429247.2011.11431892
- Yen, Y., Tseng, J., & Wang, H. (2015). The effect of internal social capital on knowledge sharing. *Knowledge Management Research & Practice*, 13(2), 214–224. https://doi.org/10.1057/kmrp.2013.43
- Yeo, R. K., & Marquardt, M. J. (2015). To share or not to share? Self-perception and knowledge-sharing intent. *Knowledge Management Research & Practice*, 13(3), 311–328. <a href="https://doi.org/10.1057/kmrp.2013.52">https://doi.org/10.1057/kmrp.2013.52</a>
- Yin, R. K. (2014). Case study research: Design and methods (5th ed.). SAGE Publications.
- Yu, Y., Hao, J., Dong, X., & Khalifa, M. (2013). A multilevel model for effects of social capital and knowledge sharing in knowledge-intensive work teams. *International*

- Journal of Information Management, 33(1), 780–790. https://doi.org/10.1016/j.ijinfomgt.2013.05.005
- Zaei, M. E., & Kapil, P. (2016, June). The role of intellectual capital in promoting knowledge management initiatives. *Knowledge Management & E-Learning*, 8(2), 317–333.
- Zaragoza-Sáez, P., Claver-Cortes, E., Marco-Lajara, B., & Ubeda-Garcia, M. (2016).

  Influence of intellectual capital upon knowledge creation in Spanish subsidiaries:

  An empirical study. *Knowledge Management Research & Practice*, 14(4), 489–501. <a href="https://doi.org/10.1057/kmrp.2015.20">https://doi.org/10.1057/kmrp.2015.20</a>
- Zyphur, M. J., & Pierides, D. C. (2017). Is quantitative research ethical? Tools for ethically practicing, evaluating, and using quantitative research. *Journal of Business Ethics*, 143(1), 1–16. <a href="https://doi.org/10.1007/s10551-017-3549-8">https://doi.org/10.1007/s10551-017-3549-8</a>

## Appendix A: Interview Questions – Managers

Participant ID Code:	Interviewer:	Robert Haughton
Title/Role:	-	
Date:	Time:	
	-	

- 1. What is your perception of knowledge retention and its importance to your project team and/or organization?
- 2. What benefit do you think knowledge retention brings to your overall project and/or organization?
- 3. As a manager, what knowledge retention strategies do you have in place to keep your team on track when there is turnover on your team?
- 4. Explain your process of replacing or substituting the lost expertise or knowledge?
- 5. Can you describe the formal and informal methods used to transfer/share knowledge? Examples of formal methods are mentorship programs or crosstraining classes. Informal methods are knowledge sharing or social networks.
- 6. What are some of the challenges you face in using the methods you just mentioned to promote knowledge retention?
- 7. How have you managed such challenges or obstacles?

## **Personal experience**

- 8. Please tell me an example when you successfully shared some of your knowledge about your project with another project manager or team lead?
- 9. Can you give me a recent example where you successfully used some information that you collected from another employee in your organization?

- 10. What type of method(s) do you use to obtain information from other managers or employees in the project/organization? In your opinion, which method(s) are the most useful?
- 11. Have there been situations when you felt that what you know (your knowledge) will be lost when you leave the project and/or the organization? If so, can you describe these situations?
- 12. Are there any other important issues concerning knowledge transfer/sharing method(s) with project team members that we have not discussed?

## Appendix B: Interview Questions – Non-managers

Partic	eipant ID Code:		Interviewer:	Robert Haughton			
Title/	Role:		-				
Date:			Time:				
1.	What is your per team and organi	•	ge retention and	d its importance to your project			
2.	2. What benefit do you think knowledge retention brings to your project/team? To your organization?						
3.	What is your perception of how your manager handles team turnover as it relates to knowledge retention?						
4.	Please tell me an example when you successfully shared some of your knowledge about the project with a new project team member?						
5.	Think about a recent task where you needed to obtain some information from other employees in the project/organization. How did you go about it?						
6.	Can you give me a recent example where you successfully used some information that you collected from another team member on your project/organization?						
7.	How important i	w important is the knowledge you have received from other team members?					
8.	What types of method(s) do you use to obtain information from other employees in the organization? In your opinion, which method(s) are the most useful?						

9. Can you give an example of a time when you obtained some information from other team members in the organization using formal method(s), such as a

training program?

- 10. Can you give an example of a time when you obtained some knowledge using informal method(s), such as social networks or knowledge sharing?
- 11. Are there any other important issues concerning knowledge transfer/sharing method(s) with other team members that we have not discussed?

## Appendix C: Email Invitation to Potential Participants

Hello,

My name is Robert Haughton, a Government Health Care consultant and a doctoral student at Walden University. I am sending this message to invite you to take part in a research study of exploring the knowledge retention strategies to help prevent knowledge loss in project-based organizations. The purpose of this study is to explore the knowledge retention strategies that help to prevent knowledge loss in project-based organizations (PBOs). I am specifically interested in the different methods managers use to retain knowledge to keep project productivity going. I am seeking individuals who fit the selection criteria below to take part in a 15-to-30-minute interview.

To be eligible for this study, you must be either:

- A manager (i.e., project coordinators, project team leaders, project managers, project directors, and senior project managers) with a minimum of 3–5 years of project management experience.
- Full-time employee (non-management) with 2–5 years of experience working on project teams
- A consultant or contractor with 2–5 years of experience working on project teams.

All participants and their organizations will remain confidential throughout this research study. To maintain the privacy of the interviews, the interview settings for the managers, employees, and consultants will occur through Skype or Zoom to allow participants to speak about their experiences handling knowledge loss and methods to retain knowledge.

If you would like to participate in this study and meet the above criteria, please email me, and I will send you the informed consent form to review and sign, along with a copy of the interview questions to review in advance.

I thank you for your consideration and your contribution towards improvement of knowledge retention strategies to prevent knowledge loss in PBOs.

Best regards,

Robert Haughton